Article

Reshaping Digital Competition: The New Platform Regulations and the Future of Modern Antitrust

The Antitrust Bulletin 2022, Vol. 67(2) 302–340 © The Author(s) 2022



Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/0003603X221082742 journals.sagepub.com/home/abx



Elias Deutscher*

Abstract

This article reflects on the way in which the new initiatives to regulate powerful online platforms in the European Union, the United States, the United Kingdom, and Germany challenge well-established fundamentals of modern antitrust and thereby reshape the future of competition law. It shows that the new platform regulations set in motion a profound transformation of modern antitrust law that operates along four parameters. First, the new platform regulations unsettle the long-standing baseline assumption that the maximization of consumer welfare constitutes competition law's core mission. Second, the new instruments repudiate the orthodox understanding of error costs that advocates under-enforcement as the optimal standard of intervention in innovation-driven markets. Third, by relying primarily on rule-like presumptions as legal commands to regulate digital competition, the new platform regulations reverse the trend toward an increasingly inductive mode of analysis that characterized modern antitrust under the "more economic" or "effects-based" approach. Fourth, the new platform regulations also fundamentally diverge from a purely probabilistic standard of proof which requires the showing that impugned conduct is more likely than not to cause anticompetitive harm. The reconfiguration of modern antitrust along these four vectors, the article concludes, foreshadows a new, more inclusive model of innovation and growth in digital markets.

Keywords

digital markets, digital platform regulations, presumptions, error-costs, standard of proof, rules, standards, consumer welfare standard, decision theory

I. Introduction

Antitrust law is set to undergo a period of tectonic shifts as policymakers in Europe and the United States are rushing toward the adoption of new regulations to tame the unprecedented economic power of digital platforms (hereinafter "new platform regulations"). While the current antitrust debate focuses primarily on the specific substantive and procedural rules, as well as the institutional design of these new regulatory tools, this contribution takes a slightly different, at the same time forward- and backward-looking angle. It approaches the new platform regulations not only as the heralds of a new era in

Corresponding Author:

Elias Deutscher, School of Law, University of East Anglia, Norwich Research Park, Norwich NR4 7TJ, UK. Email: e.deutscher@uea.ac.uk

^{*}University of East Anglia, School of Law/Centre for Competition Policy, Norwich, UK

antitrust law. Rather, as these new platform regulations considerably depart from the status quo of conventional antitrust enforcement, they also constitute a mirror that throws into relief and, thereby, allows us to better understand the basic economic and normative predispositions that shaped modern antitrust law over the last four decades. In unpacking how the new platform regulations reconfigure these predispositions, the article inquires into the broader implications that the new platform regulations may entail for the future of antitrust law. The paper argues that these new initiatives to regulate digital competition mark a "paradigm adjustment" of modern antitrust along four fault lines.

First, being geared toward promoting fairness, contestability, and non-economic values such as privacy, the new platform regulations openly reject the long-standing dogma that consumer welfare constitutes the only rational and legitimate goal of competition law. Instead, they emphasize that the protection of competition, as the key mission of competition law, extends beyond promoting consumer welfare and is capable of embracing other values.

Second, the new platform regulations also challenge the conventional wisdom that false positives of competition law enforcement are more costly than false negatives. Instead, the new platform regulations embody a recalibration of the error-cost framework which recognizes that the probability and magnitude of anticompetitive harm in digital markets may be greater than usually assumed by conventional antitrust literature.

Third, this recalibration of the error-cost framework becomes apparent in the reliance of the new platform regulations on a broad set of ex ante rules that introduce presumptions of illegality for specific types of platform conduct. By forging rule-like legal presumptions as legal commands to regulate the conduct of digital platforms, the new regulations depart from the "effects-based" analysis as the default mode of assessment of the "more economic approach." Instead of endorsing an inductive case-by-case approach, the new platform regulations highlight the value of economically informed rebuttable presumptions in antitrust analysis.

A fourth distinctive feature of the new platform regulations that also reflects the recalibration of the conventional error-cost framework is their recourse to a probabilistically de-weighted or bounded standard of proof. Instead of requiring the showing of actual or likely anticompetitive effects, the new platform regulations compel antitrust intervention on the mere basis that specific forms of conduct by powerful platforms may result in potential anticompetitive harm of significant scale. This bounded probabilism of the new platform regulations thus marks an important departure from the increasing trend in conventional antitrust analysis to make the finding of unlawful conduct conditional on the showing that anticompetitive effects are more likely than not.

Against the backdrop of this recalibration of the goals, error-cost framework, legal commands, and standard of proof, the article concludes that the new platform regulations epitomize a fundamental rethink of innovation in digital markets. In fact, the new platform regulations openly discard the Schumpeterian conception of innovation that has shaped mainstream antitrust enforcement in dynamic, high-tech markets. Instead of being concerned about the ability of large-scale incumbents to appropriate and recoup their investments in the development of innovative technology, products, and services, the new platform regulations aim to ensure market openness and contestability and preserve smaller business users' and rivals' sunk investments in digital innovation.

To illustrate how the new platform regulations disrupt and evolve competition law in readjusting the predispositions of modern antitrust, the remainder of the paper is organized as follows. Section II provides an overview of the new regulations adopted or currently discussed in Germany, the European Union (E.U.), the United Kingdom (U.K.), and the United States (U.S.) to reign in the economic power of digital platforms. Section III describes how the new platform regulations depart from the conventional, monolithic, and consumer welfare-based understanding of competition law which assumes that the ultimate mission of competition law consists of securing that consumers get a better deal in terms of lower prices, greater quality, new products, or broader choice. Section IV traces how the new platform regulations bring about a recalibration of the conventional error-cost framework of antitrust law.

Section V shows how this recalibration of the error-cost framework feeds through into the reliance of the new platform regulations on form-based, rebuttable presumptions. Section VI sheds light on how the recalibration of the error-cost framework translates into a lowered, probabilistically de-weighted or bounded standard of proof. Section VII concludes in putting this evolution into a broader context by exploring how the new platform regulations foreshadow a new model of digital innovation and growth.

II. The New Platform Regulations—An Overview

Over the last years, experts and policy makers across the world have pondered over how competition law could be reformed to tackle the challenges that the rise of digital markets poses to competition law. This process culminated in the adoption of the 10th amendment of the Competition Act in Germany,¹ as well as the proposals of a Digital Markets Act (DMA) in the E.U.,² a New Pro-Competition Regime for Digital Markets ("SMS regime") in the U.K.,³ and several legislative bills in the U.S..⁴ All initiatives have as their common aim to address growing concerns over the increasingly entrenched economic power that a handful of powerful digital platforms have amassed over the last decade.

A. The Entry-Point: The Designation of Platforms with Entrenched Substantial Market Power

Although these new platform regulations come along in different forms and shapes,⁵ they have many features in common. All new instruments revolve around a designation process that ensures that the new platform regulations only apply to the most powerful digital platforms (see Table 1). Depending on the jurisdiction at hand, these regulated platforms are referred to as "gatekeeper" platforms,⁶ platforms with "strategic market status" (SMS),⁷ "covered platform,"⁸ or "multi-sided platforms and networks holding a position of paramount significance for competition across markets."⁹ In terms of their

 A new pro-competition regime for digital markets—Advice of the Digital Markets Taskforce (hereinafter "Advice of the Digital Markets Taskforce"); A new pro-competition regime for digital markets—Consultation document July 2021. CP 489 (hereinafter "DCMS/BEIS Consultation Document").

Gesetz zur Änderung des Gesetzes gegen Wettbewerbsbeschränkungen für ein fokussiertes, proaktives und digitales Wettbewerbsrecht 4.0 und andere Bestimmungen (GWB-Digitalisierungsgesetz) (hereinafter "Competition Law 4.0") 18 January 2021, Bundesgesetzblatt Jahrgang 2021. For an English translation of the consolidated version of the 10th Amendment of the GWB Consolidated version of the 10th Amendment to the German Act against Restraints of Competition, see *German Competition Act 2021—Unofficial Translation*, https://www.d-kart.de/wp-content/uploads/2021/01/GWB-2021-01-14-engl.pdf.

Proposal for a Regulation of the European Parliament and of the Council on contestable and fair markets in the digital sector (Digital Markets Act) (hereinafter "DMA Proposal"). COM/2020/842 final.

H.R.3816—American Choice and Innovation Online Act (hereinafter "ACIO Act") 2021, 117th Congress (2021–2022); H.R.3825—Ending Platform Monopolies Act (hereinafter "EPM Act") 2021, 117th Congress (2021–2022); H.R.3826— Platform Competition and Opportunity Act (hereinafter "PCO Act") 2021, 117th Congress (2021–2022); H.R.3849— ACCESS Act (hereinafter "ACCESS Act") 2021, 117th Congress (2021–2022). H.R.3843—Merger Filing Fee Modernization Act (hereinafter "MFFM Act)", 2021, 117th Congress (2021–2022).

^{5.} For instance, the U.S. proposals foresee an important role of the U.S. court system in enforcing the new rules, whereas the enforcement of the E.U., U.K., and German rules is largely within the competence of administrative authorities. Monika Schnitzer et al., *International Coherence in Digital Platform Regulation: An Economic Perspective on the US and EU Proposals* (2021), at 12, https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3923604.

^{6.} DMA Proposal, *supra* note 2, art. 3 (1).

^{7.} Advice of the Digital Markets Taskforce, *supra* note 3, paras. 4.7–4.12.

^{8.} ACCESS Act, *supra* note 4, s. 5 (6); EPM Act, *supra* note 4, s. 5 (4); PCO Act, *supra* note 4, s. 3 (d); ACIO Act, *supra* note 4, s. 2 (d) and (g) (4).

^{9.} Competition Law 4.0, supra note 1, s. 19a (1) in conjunction with s. 18 (3a).

| | U.S. Proposals | U.K. New Pro-Competition Regime for Digital Markets | E.U. Digital Markets Act Proposal | German Competition Law 4.0 |
|----------------------------|--|---|---|--|
| Designating authority | Bureau of Digital Markets within the Federal Trade Commission and Antitrust Division of the Department of Justice | Digital Markets Unit (DMU) | European Commission | German Federal Cartel Office (FCO) |
| Addressee | Covered platforms | Platforms with strategic market status (SMS) | Gatekeeper platforms active in core platform services | Platforms with paramount significance for competition across markets |
| Quantitative criteria | s. 5 (6) ACCESS Act s. (5) (4) EPM Act s. 3 (d) PCO Act s. 2 (d) and 2 (g) (4) ACIO Act | Only for purposes of prioritization of specific platforms / activities | Art. 3 (2) and (3) | None |
| Qualitative criteria | s. 5 (8) ACCESS Act s 5 (6) EPM Act s. 3 (f) PCO Act s. 2 (g) (6) ACIO Act | Advice of the Digital Markets Taskforce, paras. 4.4, 4.9–4.24 DCMS/BEIS paras. 60–78 | Art. 3 (6) | s. 19a (1) and s. 18 (3a) |
| Designation | Presumption Investigation and ex officio designation | Investigation and ex officio designation | Rebuttable presumption and notification Investigation and ex officio designation | Investigation and ex officio designation |
| Duration of designation | 10 years | 5 years | 2 years | 5 years |

Table 1. Designation Process and Criteria.

material scope, the new regulations apply to sectors or activities for which digital technology constitutes a core element of their functioning and business model. Whereas some of the new regulations only require that the regulated platforms be active in loosely defined digital markets or sectors,¹⁰ others exclusively apply to platforms that operate a specific subset of clearly delineated digital activities.¹¹

The designation of regulated platforms proceeds through two main channels. Regulated platforms are designated

- either on the basis of specific quantitative thresholds—such as annual turnover, number of end users and business users, market capitalization—that provide proxies for their size, competitive impact, and market power;
- or based on qualitative criteria—such as entry barriers, network effects, barriers to switching, dominance—which are indicative of their substantial and enduring market power.

Some jurisdictions, such as the E.U. and the U.S., rely on a combination of both quantitative and qualitative scoping criteria. The U.K. and Germany only use qualitative criteria to designate regulated platforms. Whereas in Germany and the U.K. platforms are called in by competition authorities/regulators ex officio after a case-specific analysis, the quantitative criteria in the E.U. DMA trigger a rebuttable presumption that a platform holds gatekeeper status.¹² Accordingly, a provider of a core platform service that meets the quantitative thresholds set out in the DMA is presumed to qualify as a designated platform and must notify the Commission about its status, unless it puts forward countervailing evidence showing that it does not possess gatekeeper power.¹³ The U.S. proposals provide for a similar presumption for platforms that meet the primarily quantitative designation criteria.¹⁴ Designated platforms are subject to the new platform regulations for a duration between 2 (E.U.), 5 (Germany and the U.K), and 10 (the U.S.) years.

The designation process of platforms under the new platform regulations importantly differs from traditional antitrust analysis in two respects. First, although the designation process requires competition authorities to identify specific industries in which the platforms are active, they are no longer under a formal obligation to define relevant antitrust markets. Second, while the scoping criteria are geared toward identifying platforms with substantial and enduring market power, competition authorities no longer need to make any formal finding that the platforms hold a dominant or monopoly position. Though there might be an overlap between the status of designated platforms and dominance/monopoly power, they do not always have to be congruent. This becomes, for instance, apparent in the fact that the DMA and the German Competition Law 4.0 also apply to platforms that have not yet reached a dominant position.¹⁵ This extension of the regulation of unilateral conduct beyond dominance/monopoly power is testament to the growing concern that the traditional concepts of "dominance" or "monopoly" fail to account for the fact that digital platforms may wield important "intermediation

See for instance ACIO Act, *supra* note 4, s. 2 (g) (10). Competition Law 4.0, *supra* note 1, s. 19a (1) in conjunction with s. 18 (3a). For a discussion of the scope of s. 19a and its potential to cover firms which are not digital platforms in the conventional sense, Jens-Uwe Franck & Martin Peitz, *Digital Platforms and the New 19a Tool in the German Competition Act*, 12 J. EUR. COMPET. LAW PRACT. 513, 517–19 (2021); DCMS/BEIS Consultation Document, *supra* note 3, paras. 55–57.

^{11.} The DMA applies to so-called core platform services that play a crucial role in matching business and end users: namely, online intermediation services, online search engines, social networking services, video-sharing platforms services, number-independent interpersonal communication services, operating systems, cloud computing, and advertising services. DMA Proposal, *supra* note 2, art. 2 (2) (a) to (h).

^{12.} Id. art. 3 (2) and (3).

^{13.} Id. art. 3 (2) to (4).

^{14.} PCO Act, supra note 4, s. 3 (d); ACIO Act, supra note 4, s. 2 (d) and (g). Schnitzer et al., supra note 5, at 9.

^{15.} DMA Proposal, *supra* note 2, art. 3 (1) (c), art. 3 (6) and art. 15 (4). The new German Competition Law 4.0 also explicitly covers platforms that are not yet dominant Competition Law 4.0, *supra* note 1, s. 19a (2) 3.

power" allowing them to harm competition, even though they do not hold a dominant position in a clearly defined relevant product market.¹⁶

B. Ex Ante Conduct and Merger Rules

A second common feature of the new platform regulations is that they subject designated platforms to a broad set of ex ante rules and obligations. The peculiarity of these rules is that they are—at least to a certain degree—self-executing. This means that these rules are designed in a way that reduces the need for any further specification to a minimum. Instead, most of the ex ante rules are directly applicable and binding on the designated platforms.

The specific content and design of the ex ante rules differ across the various new instruments. Whereas the E.U., U.S., and German platform regulations crafted a set of ex ante rules that apply to all designated platforms across the board, the new U.K. regime for SMS platforms envisages a more bespoke approach that lays down legally binding codes of conduct for individual designated platforms. These codes blend some ex ante rules that apply to all designated platforms with more tailored rules that are negotiated with the designated platforms on an individual basis.¹⁷

Despite these differences in their design, all platform regulations share a preventative approach in so far as they seek to bar designated platforms from engaging in certain forms of conduct by outlawing them ex ante. There is also some considerable overlap in terms of the substantive content of these ex ante rules across the different platform regulations (see Table 2):

- Data: All platform regulations provide for rules that regulate the extent to which regulated platforms can make use of and must provide third-party access to data. Most platform regulations restrict the ways in which designated platforms can combine data sourced from their different services or make use of non-public data of business users to gain a competitive edge over the latter. They also contain obligations to enable and/or prohibitions to impede data portability for end users and/or business users. Moreover, they mandate designated platforms to ensure that business users can access data generated by them or their end users on the platform. The DMA even goes one step further by imposing an obligation on gatekeeper platforms operating search engines to give competing search engine operators access to their data.¹⁸
- Interoperability and switching: Prohibitions to restrict and obligations to ensure interoperability between the designated platform's products or services and the products and services provided by third parties are also present in all new platform regulations. Furthermore, the new platform regulations lay down rules that promote switching and multi-homing.
- Tying: Designated platforms are also prohibited from using contractual or technical ties, and in certain circumstances default settings, to give their own service a competitive edge over those of competing third parties.
- Self-preferencing: All new platform regulations outlaw self-preferencing of different sorts, whereby a vertically integrated platform affords preferential treatment, visibility, or ranking to its own services, while placing competing rivals at a competitive disadvantage.

Heike Schweitzer et al., Modernisierung der Missbrauchsaufsicht für marktmächtige Unternehmen, PROJEKT IM AUFTRAG DES BUNDESMINISTERIUMS FÜR WIRTSCHAFT UND ENERGIE (BMWi), (2018), at 40–42, https://www.bmwi.de/ Redaktion/DE/Publikationen/Wirtschaft/modernisierung-der-missbrauchsaufsicht-fuer-marktmaechtige-unternehmen. pdf?__blob=,publicationFile&v=15.

^{17.} Advice of the Digital Markets Taskforce, *supra* note 3, paras. 4.33–4.44. DCMS/BEIS Consultation Document, *supra* note 3, paras. 86–91.

^{18.} Along similar lines, the 10th amendment of the German Competition Act introduces a legal basis for competitors to obtain access to the data held by dominant firms, by explicitly recognizing that a refusal to grant data access may amount to an abuse of dominance. Competition Law 4.0, *supra* note 1, s. 19 (2)4.

| Prohibitions/Obligations | U.S. Proposals | U.K. New Pro-Competition Regime for Digital Markets ¹⁹ | E.U. Digital Markets Act Proposal | German Competition Law 4.0 |
|---|--|---|--|---|
| Cross-usage of data collected through different proprietary/ third party services | | | Art. 5 (a) | s. 19a (2) 4 (a) and (b) |
| Use of non-public data sourced from other business users active on the platform | s. 2 (b) (3) ACIO Act | | Art. 6 (1) (a) | s. 19a (2) 4 (b) |
| Data portability | s. 2 (b) (4) ACIO Act (business users) s. 3 ACCESS Act (business and end users) | | Art. 6 (1) (h) (business and end users) | s. 19a (2) 5 |
| Access to data generated by users | s. 2 (b) (4) ACIO Act (business users) | | Art. 6 (1) (i) (business and end users) | |
| Data sharing with competitors | | | Art. 6 (1) (j)) for third party providers of online search engines | s 19 (2) 4 refusal to grant access to data |
| Rebalancing of asymmetric bargaining power | s. 4 (f) ACCESS Act (Data minimization) | Principles I(a), 3 (a) to (c) | Art. 5 (g) and Art. 6 (1) (g) | s. 19a (2) 6 and 7 |
| Self-preferencing Price-parity or Most Favored Nation (MFN) clauses | s. 2 (a) (1)–(2) and s. 2 (b) (7) ACIO Act s. 2 (b) (8) ACIO Act (narrow and wide MFNs) | Principle 2 (a) | Art. 6 (I) (d) Art. 5 (b) (only wide MFNs) | s. 19a (2) 1 (a) |
| Anti-steering Tying and defaults | s. 2 (b) (6) ACIO Act s. 2 (b) (2) and (5) ACIO Act | Principles I (c) and 2(d) Principle 2 (b) | Art. 5 (c) Art. 5 (e) and (f) Art. 6 (1) (b) and (c) | s. 19a (2) 2 (b) s. 19a (2) 1 (b), 2 (a), and 3 |
| (Secondary-line) discrimination | s. 2 (a) (3) ACIO Act | Principle I (b) | Art. 6 (1) (k) (Access for business users to software | |
| Interoperability | s. 4 ACCESS Act | | Art. 6 (f) | s.19a (2) 5 |
| | | | | |

Table 2. Ex Ante Rules in the New Platform Regulations.

^{19.} DCMS/BEIS Consultation Document, supra note 3, para. 82 and Figure 4.

| | U.S. Proposals | U.K. New Pro- Competition Regime for Digital Markets | E.U. Digital Markets Act Proposal | German Competition Law 4.0 |
|--------------------------|--|---|---|---|
| Platform acquisitions | Prohibition unless proof that they do not harm (nascent) competition s. 2 (a) and (b) PCO Act | Notification obligation and lowering of the standard of proof | Notification obligation (Art. 12) | None, but transaction value (400m EUR) threshold introduced in the 9th amendment of the German Competition Act in 2017 |

Table 3. Merger Rules in the New Platform Regulations.

- Discrimination: Other forms of discrimination against third-party business users are also addressed by most platform regulations. Designated platforms thus are—at least to a certain degree—subject to a public-utility like obligation to guarantee fair, reasonable and non-discriminatory access to third parties on their platforms. Whereas the E.U., U.K., and German regulations translate this non-discrimination principle primarily through narrowly defined prohibitions and obligations, the U.S. proposals also stipulate broadly phrased general prohibitions of discriminatory conduct.²⁰
- Anti-steering provisions and Most Favored Nation (MFN) clauses: A majority of the new platform regulations also prohibit contractual or technical anti-steering provisions and, albeit to a lesser extent, wide and narrow MFN clauses. They thereby seek to prevent designated platforms from restricing the ability of third-party business users to reach end users through alternative sales channels.
- Bargaining power: In addition, most new platform regulations also contain a number of rules that seek to address the bargaining power of designated platforms, for instance, by promoting greater price and service transparency for business and end users, notably in opaque markets such as the online advertising sector.²¹

Unlike in regular antitrust cases, designated platforms have only limited possibilities to obtain an exemption from or justify violations of these obligations. The U.S. proposals, DMA proposal, and German Competition Law 4.0 do not provide for any explicit efficiency defense. Instead, designated platforms can only advance a narrowly construed "affirmative defense,"²² "objective justifications,"²³ "economic viability,"²⁴ or "public policy"²⁵ grounds to excuse any non-compliance with the newly designed rules. Only the U.K. framework explicitly recognizes the possibility for firms to plead an efficiency defense to excuse conduct that otherwise violates the code of conduct.²⁶

Alongside ex ante rules relating to unilateral conduct of powerful platforms, new platform regulations also contain provisions that seek to empower competition authorities to scrutinize start-up acquisitions by powerful platforms (see Table 3). While the DMA only imposes a new obligation on gatekeeper

^{20.} ACIO Act, supra note 4, s. 2 (a) (1) to (3); Schnitzer et al., supra note 5, at 15.

^{21.} Competition and Markets Authority (CMA), *Online Platforms and Digital Advertising: Market Study Final Report* (2020), https://assets.publishing.service.gov.uk/media/5efc57ed3a6f4023d242ed56/Final_report_1_July_2020_.pdf.

^{22.} ACIO Act, supra note 4, s. 2 (c).

Beschlussempfehlung und Bericht des Ausschusses f
ür Wirtschaft und Energie (9. Ausschuss)—Entwurf eines Gesetzes zur Änderung des Gesetzes gegen Wettbewerbsbeschr
änkungen f
ür ein fokussiertes, proaktives und digitales Wettbewerbsrecht 4.0 und anderer wettbewerbsrechtlicher Bestimmungen (GWB-Digitalisierungsgesetz) 13 January 2021. Drucksache 19/25868 113, 116–17; Franck & Peitz, *supra* note 10, at 521–22.

^{24.} DMA Proposal, *supra* note 2, art. 8.

^{25.} Id. art. 9.

^{26.} Advice of the Digital Markets Taskforce, supra note 3, para. 4.40.

platforms to notify any acquisition to the European Commission,²⁷ the new U.K. SMS regime also envisages a lowering of the standard of proof for the competitive assessment of mergers by designated digital platforms. When reviewing acquisitions by platforms with SMS status in phase II, the Competition and Markets Authority (CMA) would no longer have to establish that the merger will lead on a "balance of probabilities" to a substantial lessening of competition (SLC) as is the case for regular mergers. Instead, it could rely on the lower "realistic prospect" standard, normally reserved for the phase I assessment of mergers. Unter this lower standard, it only needs to demonstrate that the merger gives rise to a realistic prospect of a SLC.²⁸

The U.S. proposals go even further in altering the assessment of mergers by powerful digital platforms. They envisage a presumption of unlawfulness against acquisitions by designated platforms. Under the new platform rules, designated platforms will be prohibited from acquiring other firms unless they advance "clear and convincing evidence" showing that the merger would not adversely affect competition.²⁹

The German Competition Law 4.0 does not provide for specific rules on start-up acquisitions. However, already back in 2017, the 9th Amendment to the German Competition Act had introduced a new transaction value threshold (s. 35(1a) 3) to enable the Federal Cartel Office to review start-up acquisitions that otherwise would fly under the radar of the jurisdictional turnover thresholds.³⁰

C. New Enforcement Tools

A third feature which all new platform regulations have in common is that they empower competition authorities with a broad range of enforcement tools and remedies to ensure compliance with the new obligations and enhance competition in digital markets (Table 4). All platform regulations provide for interim or emergency relief measures aimed at speeding up enforcement in digital markets. In the event of non-compliance, designated platforms will face substantial fines. The self-executing nature of most prohibitions and rules in the new platform regulations also entails that they clearly indicate a specific behavioral remedy that platforms have to adopt in order to bring an infringement to an end. Alongside behavioral remedies, the new platform regulations also provide for a broad range of structural remedies. They, however, differ in the way in which they blend and escalate these different remedies.

The U.S. proposals, for instance, go as far as envisaging lines of business restrictions that prohibit platforms from operating lines of business through which they provide services or sell goods via their platform. Designated platforms may also be required to divest existing lines of business if they create conflicts of interest. These structural remedies may apply to designated platforms even in the absence of any concrete breach of the new platform regulations. The U.S. proposals thus create a non-fault antitrust liability for designated platforms that exposes them to structural interventions in markets whose characteristics make systemic competition issues more likely.³¹

The U.K. SMS framework also introduces a similar non-fault liability regime for designated digital platforms. The SMS regime, on one hand, provides for the imposition of remedies to address non-compliance with the new codes of conduct. On the other, it also empowers the newly created Digital Markets Unit (DMU) to adopt pro-competition interventions (PCIs) that impose behavioral and structural remedies on designated platforms with a view to enhancing the contestability of digital markets. These PCIs can be

^{27.} DMA Proposal, supra note 2, art. 12.

^{28.} Advice of the Digital Markets Taskforce, *supra* note 3, paras. 4.149 - 4.153 and Annex F paras. 89–132.

^{29.} PCO Act, supra note 4, s. 2 (a) and (b).

Gesetzentwurf der Bundesregierung—Entwurf eines Neunten Gesetzes zur Änderung des Gesetzes gegen Wettbewerbsbeschränkungen 7 November 2016 39; Neuntes Gesetz zur Änderung des Gesetzes gegen Wettbewerbsbeschränkungen 8 June 2017. Bundesgesetzblatt Jahrgang 2017 Teil I Nr. 33 s. 35 (1a) 3.

Such a non-fault line of business restriction is proposed in the EPM Act, *supra* note 4, s. 2 (a); If violations of the ACIO Act arise from conflicts of interests, similar line of business restrictions can be applied. ACIO Act, *supra* note 4, s. 2 (f) (2) (D); Schnitzer et al., *supra* note 5, at 24–25.

| Type of Intervention | U.S. Proposals | U.K. New Pro-Competition Regime for Digital Markets (SMS Regime) | E.U. Digital Markets Act Proposal | German Competition Law 4.0 |
|----------------------------|---|---|---|----------------------------------|
| Fines | s. 10 (a) ACCESS Act s. 2 (f) (1) ACIO Act s. 3 (c) EPM Act | Advice of the Digital Markets Taskforce paras. 4.86; 4.96 DCMS/BEIS para. 127 | Art. 26 (1) Art. 27 | Existing fines |
| Behavioral remedies | s. 10 (b) ACCESS Act s. 2 (f) (2) ACIO Act s. 5 (f) PCO Act s. 4 EPM Act | Advice of the Digital Markets Taskforce paras. 4.55, 4.63; 4.90 DCMS/BEIS paras. 93, 99 | Art. 7 Art. 11 Art. 25 (3) Art. 16 (1) | Existing remedies |
| Structural remedies | s. 2(f)(2) (D) AICO Act s. 2 (a) EPM Act s. 5 (f) PCO Act | Pro-competition interventions (PCIs) Advice of the Digital Markets Taskforce paras. 4.68 – 4.69; DCMS/BEIS para. 111 | Art. 16 (2) | Existing remedies |
| Interim measures | s. 9 (b) ACCESS Act s. 2 (i) ACIO Act s. 5 (e) PCO Act | Advice of the Digital Markets Taskforce paras. 4.86, 4.98; BEIS/DCMS; para. 100 | Art. 22 | s. 32a |
| Commitment decisions | | | Art. 23 | Existing commitment procedure |
| Non-fault interventions | Line of business restrictions s. 2 (a) EPM Act | PCIs Advice of the Digital Markets Taskforce paras. 4.60-4.81 DCMS/BEIS paras. 102-123 | Market investigation tool and update of core platform services and obligations Arts. 10 and 17 | |
| | | | | |

Table 4. Enforcement Tools.

mandated regardless of whether a SMS platform has breached its code of conduct. Instead, they are explicitly aimed at eliminating the root causes of the entrenched market power of SMS platforms.³²

Similarly, the DMA proposal gives the Commission at hand a new market investigation tool which empowers it to go beyond the enforcement of the rules and obligations laid down in the DMA. Instead, the Commission can use the market investigation tool to identify new digital sectors in which markets are not sufficiently contestable or determine new practices that unfairly hinder competition or dampen contestability.³³ These markets or practices can then be added to the list of "core platform services" regulated³⁴ and of blacklisted practices outlawed by the DMA.³⁵ In a similar vein as the non-fault liability regimes in the U.S. and U.K. proposals, the market investigation tool enables the Commission to intervene in markets and impose obligations on digital platforms that are not geared toward remedying specific breaches of existing competition and platform rules but—in their final consequence—seek to reduce entrenched market concentration as such.

In sum, the new platform regulations in Germany, the E.U., the U.K., and the U.S. bring about a fundamental reconfiguration of the regulation of competition in the digital economy. Despite some differences in their institutional and substantive design, all regulatory initiatives display a number of common features (Tables 1 to 4). In terms of material scope, all four have in common that they apply to multi-sided platforms in digital markets. For a digital platform to fall within the scope of the new regulations it has to hold some form of durable, significant market power with respect to digital markets that play a crucial role as access points for businesses to reach end consumers. Platforms with substantial and durable market power will be identified on the basis of quantitative (E.U., the U.S.) and qualitative (Germany, E.U., the U.S., the U.K.) factors. Competition authorities are, however, under no obligation to define standard antitrust markets to determine the market power of digital platforms. Moreover, platforms may become subject to the new regulatory framework even if their market power falls short of conventional dominance or monopolization thresholds. All four new platform regulations also have in common that they lay down a number of ex ante rules that prohibit specific forms of conduct deemed to undermine the contestability of digital markets and/or amount to unfair conduct toward business or end users. Most new platform regulations also strengthen the ability of competition authorities to scrutinize mergers by designated platforms. Beyond that, the new frameworks confer upon competition authorities far-reaching powers to ensure compliance with these newly created rules and intervene in digital markets without finding concrete unlawful behavior.

D. The Economic Rationale Underpinning the New Platform Regulations

This reconfiguration of competition law envisaged by the new platform regulations in Germany, the E.U., the U.K., and the U.S. constitutes in the first place a response to the specific features of digital markets. The most distinctive characteristic of these markets is the presence of firms that operate multi-sided online platforms. The business model of these multi-sided platforms consists of reducing transaction costs by bringing together and matching the supply and demand of end user and business user groups which value each other.³⁶ Owing to the importance of multi-sided platforms acting as intermediaries between different customer groups, digital markets are characterized by significant direct and indirect network effects.³⁷ The direct and (cross-platform) indirect network effects are further compounded by the role data play in the business model of online platforms.³⁸ As the ability of a

^{32.} Advice of the Digital Markets Taskforce, supra note 3, paras. 4.60-4.81.

^{33.} DMA Proposal, *supra* note 2, art. 17.

^{34.} Id. art. 2.

^{35.} Id. arts. 10, 17.

^{36.} Jean-Charles Rochet & Jean Tirole, Platform Competition in Two-Sided Markets, 1 J. EUR. ECON. ASSOC. 990 (2003).

^{37.} Michael L. Katz & Carl Shapiro, Network Externalities, Competition, and Compatibility, 75 Am. ECON. REV. 424 (1985).

Daniel L. Rubinfeld & Michal S. Gal, Access Barriers to Big Data, 59 ARIZ. L. REV. 339, 355–56 (2017); MAURICE E. STUCKE & ALLEN P. GRUNES, BIG DATA AND COMPETITION POLICY 23, 45, 200–205 (Oxford University Press 2016).

platform to collect data and optimize its intermediation services increases with the size of its end user base, data tend to further reinforce the direct and indirect network effects on both sides of the platform. The role of data thus amplifies already existing direct and indirect network effects because access to a broad scale and scope of data enables platforms to offer better targeted and more bespoke products and services to end and business users.

This prevalence of direct, indirect, as well as data-driven network effects has three implications for competition in digital markets. First, digital markets are often characterized by extreme economies of and returns on scale and scope. In order to be attractive for business and end users and be able to compete effectively, new platform entrants must achieve a minimum efficient scale in terms of the size and scope of their customer base, data sets, and network effects. User- and data-driven network effects thus often constitute important barriers to entry for new platform competitors. Digital markets may, therefore, only accommodate a limited number of players large enough to operate at a minimum efficient scale.³⁹ Second, data- and non-data-driven network effects may generate important lock-in effects if they make it more costly for end and business users to switch to competing platforms. This is particularly the case if network-effects-driven switching costs limit or prevent multi-homing.⁴⁰ Third, by virtue of the important role of network and lock-in effects, digital markets may reach a point where the entire demand "tips" toward a single winner that attracts so many users that it will virtually become the sole viable system in the market. A player who succeeds in achieving scale and harnessing network effects may thus be able to tilt the market on a lasting basis in its favor.⁴¹ The market position of the successful survivor of this "winner takes most" competition is often difficult to challenge and dislodge by new entrants or residual competitors. The combined effect of these extreme economies of scale and scope, network effects, and market tipping is an important driver of market concentration and decreasing contestability of digital markets. Numerous digital markets are, therefore, showing signs of increasing consolidation and concentration of economic power in the hands of a few large-scale incumbents who managed to entrench their market position.

A third distinctive feature of digital markets is the prevalence of vertical integration and the emergence of conglomerate "ecosystems."⁴² Many digital platforms integrate vertically or "diagonally" by providing up-stream, down-stream, or complementary services and expanding their presence into neighboring markets. As a consequence, digital platforms often rely on a hybrid business model.⁴³ On one hand, they operate a marketplace facilitating the interaction between business users and end users. In providing this marketplace function, digital platforms control an important "input" or "bottleneck"⁴⁴ that business users have to access to reach end users. On the other hand, digital platforms often also operate a "sales function" through which they distribute their own complementary products or services on their marketplace.⁴⁵

Jacques Crémer et al., Competition Policy for the Digital Era (2019), at 19–24, https://op.europa.eu/en/publication-detail/-/publication/21dc175c-7b76-11e9-9f05-01aa75ed71a1/language-en; Jason Furman et al., Unlocking Digital Competition (2019), paras. 1.68–1.70, https://www.hks.harvard.edu/centers/mrcbg/programs/growthpolicy/ unlocking-digital-competition-report-digital-competition-expert.

^{40.} CARL SHAPIRO & HAL R. VARIAN, INFORMATION RULES: A STRATEGIC GUIDE TO THE NETWORK ECONOMY 104 and Chapter 5 (Harvard Business School Press 1999); Furman et al., *supra* note 39, paras. 1.81–1.88.

^{41.} Michael L. Katz & Carl Shapiro, *Systems Competition and Network Effects*, 8 J. ECON. PERSPECT. 93, 106 (1994); Furman et al., *supra* note 39, para. 2.5.

^{42.} Crémer et al., *supra* note 39, at 33–35. Michael G. Jacobides et al., *Toward a Theory of Ecosystems*, 39 STRAT. MGMT. J. 2255 (2018).

Simon P. Anderson & Özlem Bedre-Defolie, *Hybrid Platform Model*, CEPR Discussion Paper DP16243 (2021), https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3886686.

Thomas G. Krattenmaker & Steven C. Salop, Anticompetitive Exclusion: Raising Rivals' Costs to Achieve Power over Price, 96 YALE L.J. 209, 226–27 (1986).

^{45.} Anderson & Bedre-Defolie, supra note 43, at 2.

These three structural features of digital markets—namely, the (1) importance of multi-sided intermediary platforms, the (2) prevalence of extreme economies of scale and scope, network effects and market tipping, and (3) the presence of vertical/diagonal integration—importantly shape the incentives of incumbent digital platforms both with respect to their horizontal relationship with other (competing) platforms and with respect to their non-horizontal relationship with business users.

When it comes to the horizontal relationship between platforms, the importance of extreme returns to scale and scope and network effects increases the incentives of incumbent digital platforms to tip markets on a lasting or even irreversible basis. Once a market has pivoted in its favor, a digital platform can entrench its market power and generate monopoly profits through various monetization channels without having to fear that it will be challenged by new or competing platforms. The propensity of digital markets to tip also heightens the incentive of incumbents to defend their entrenched market position by nipping competitive threats of nascent platforms or entrants in the bud. The payoffs that an incumbent who successfully unlevels the market in its favor derives from securing a lasting monopoly position oftentimes exceed those for competitors to enter. Due to these "asymmetric stakes," an incumbent platform may indeed be inclined to to spend more (or to sacrifice more profits) to insulate its monopoly profits from potential competition than competitors or entrants may be willing to invest in order to remain operational or achieve a viable competitive to estimate horizontal competitive threats by engaging in exclusionary conduct or acquiring them.

In terms of the vertical relationship between platforms and business users, the hybrid business model of operating a "marketplace" and "sales function" often exposes digital platforms to a conflict of interest. Hybrid platforms provide business users with an important input, while at the same time competing with them downstream in selling complements to end users. Digital platforms thus face a continuous trade-off between wholesale profits that they earn by charging business users for their marketplace function and the cannibalization of the margins of their retail sales function.⁴⁷ On one hand, the more business users they serve with intermediary services, the higher their wholesale profits (wholesale margin effect). On the other hand, the more business users populate their platforms, the greater the downstream competition that "cannibalises" the profitability of their retail sales function (cannibalization effect). If the cannibalization effect outweighs the wholesale margin effect, platforms have an incentive to raise business users' (i.e., their downstream rivals') costs, for instance, by designing the rules of their market function in a way which places competing business users at a disadvantage or by leveraging their market power from the marketplace into the sales function.

These horizontal and vertical effects do not operate in isolation but often interact and reinforce each other. The various platform regulations adopted or currently envisaged by policy makers across the globe seek to address and channel the complex incentive structure of digital platforms shaped by the three effects outlined above. All types of ex ante rules and interventions foreseen in the German, E.U., U.K., and U.S. frameworks for platforms can be categorized as either (1) seeking to preserve or inject horizontal *inter*-platform competition and advance contestability, or (2) guarantee *intra*-platform competition by protecting business users or end users which are dependent on a platform's marketplace function from opportunistic behavior, or (3) both. The regulatory rationale of the new platform regulations thus mirrors the multi-dimensional harms that may derive from the horizontal and vertical incentive structure digital platforms are subject to. On one hand, by promoting contestability the platform regulations seek primarily to reduce structural factors that facilitate the entrenchment of incumbents' market power and mute their incentives to impair horizontal competition by foreclosing or acquiring competitors. On the other, by ensuring fairness and equality of opportunity, the platform regulations seek to reduce the vertical competition issues arising

^{46.} Krattenmaker & Salop, *supra* note 44, at 269–70, fn 190. Erik Hovenkamp & Steven C. Salop, *Asymmetric Stakes in Antitrust Litigation*, USC Legal Studies Research Papers Series No. 20-12 (2020), 1, https://papers.scm/sol3/papers.cfm?abstract_id=3563843.

^{47.} Anderson & Bedre-Defolie, *supra* note 43, at 16–23, 35–36.

from the hybrid role of integrated platforms that operate a simultaneous marketplace and sales function. To this end, the various platform regulations lay down a number of—primarily negative—obligations that are geared toward tempering the incentives of platforms to raise the costs of business users that act as third-party sellers competing with their own sales function on their platform.

III. Goals Reconsidered

The fact that the new platform regulations pursue the goals of enhancing contestability and fairness is a first point in which they radically divert from the baseline approach of modern antitrust. The new platform regulations indeed depart from the fundamental prior of modern antitrust that competition law has as its primary or even exclusive mission the enhancement of consumer welfare in the form of lower prices, greater choice, quality, or innovation.⁴⁸

A. The Multi-Value Approach of the New Platform Regulations

Instead of focusing on consumer welfare as their primary or exclusive goal, the new platform regulations identify several policy goals. By way of example, the proposed DMA pursues the twin goal of ensuring the contestability⁴⁹ of digital markets and fairness and equality of opportunity⁵⁰ by reducing "significant dependencies"⁵¹ of business users on powerful platforms and redressing power imbalances.⁵² Along similar lines, the U.S. proposals pursue the goal of enhancing fairness,⁵³ business opportunities,⁵⁴ choice, and innovation⁵⁵ in digital markets. The U.K. framework, too, deviates from a purely consumer welfare approach, as it seeks to address the adverse economic but also societal implications of the accumulation of market power by a small number of large-scale platforms.⁵⁶ The objectives of limiting and securing the contestability of economic power of digital platforms and at the same time guaranteeing fair and equal competitive opportunities for rivals and business users who are dependent on platform services also lie at the heart of the new German Competition Law 4.0.⁵⁷

The prominent role of contestability and fairness as goals of the new platform regulations marks a significant move away from the single-edged understanding of competition law advocated by the proponents of the consumer welfare standard. This becomes most apparent in the fact that the new platform regulations draw up ex ante rules that seek to prevent platforms from foreclosing competitors or business users regardless of their efficiency. The new platform regulations allow competition authorities to intervene in digital markets without being required to carry out an "as-efficient competitor test" which has been widely endorsed in modern U.S.⁵⁸ and E.U.⁵⁹ antitrust as the ultimate touchstone to

^{48.} HERBERT HOVENKAMP, THE ANTITRUST ENTERPRISE: PRINCIPLE AND EXECUTION 1 (Harvard University Press 2005).

^{49.} DMA Proposal, supra note 2, art. 1 (1), recitals 3-6, 10.

^{50.} Id. art. 1 (1), recitals 2-5, 10; Explanatory Memorandum, at 10.

^{51.} Id. Explanatory Memorandum, at 1.

^{52.} Id. recital 4.

^{53.} See for instance ACCESS Act, supra note 4, s. 6 (c); EPM Act, supra note 4, at 11.

^{54.} PCO Act, *supra* note 4, s.1.

^{55.} ACIO Act, supra note 4, s. 1.

^{56.} Advice of the Digital Markets Taskforce, *supra* note 3, paras. 4, 2.7–2.10, 3.5. The U.K. Government seems to favor a narrower notion of competitive harm, see DCMS/BEIS Consultation Document, *supra* note 3, paras. 28–31.

^{57.} Gesetzentwurf der Bundesregierung. Entwurf eines Gesetzes zur Änderung des Gesetzes gegen Wettbewerbsbeschränkungen für ein fokussiertes, proaktives und digitales Wettbewerbsrecht 4.0 und anderer wettbewerbsrechtlicher Bestimmungen (GWB-Digitalisierungsgesetz) 19 October 2020. Drucksache 19/23492 73, 75.

Brooke Group v. Brown & Williamson Tobacco Corp. 509 U.S. 209 (1993) 223–224; Weyerhaeuser Co. v. Ross-Simmons Hardwood Lumber Co. 549 U.S. 312 (2007) 325.

^{59.} Case C-62/86 AKZO v Commission ECLI:EU:C:1991:286 para. 71; Case C-209/10 Post Danmark A/S v Konkurrencerådet ECLI:EU:C:2012:172 paras. 22, 25. See, however, for a recent limitation of the application of the as-efficient competitor test to price conduct Case T-612/17 Google and Alphabet v Commission (Google Shopping) ECLI:EU:T:2021:763 para. 538.

determine when impugned unilateral conduct harms consumer welfare and therefore falls afoul of competition rules. The new platform regulations hence clearly depart from the fundamental of modern antitrust that unilateral conduct by dominant firms is only objectionable if it forecloses an "as-efficient" or equally efficient competitor.⁶⁰

The new platform regulations thus enable competition authorities to protect a competitive market structure and the economic opportunity and liberty of smaller competitors and business users regardless of their efficiency and immediate contribution to consumer welfare. They thereby implicitly challenge the truism of modern antitrust that competition law protects "competition, not competitors."⁶¹ In so doing, they vindicate the long-standing insight that competition, at least in the medium-to-long-run, "requires the existence of competitors, in the plural."⁶² This suggests that for competition to exist and thrive, competitors must at least be protected to some extent. For without competitors, there is no competition.

B. Toward a More Holistic Understanding of Consumer Welfare

Does this mean that the new platform regulations discard the consumer welfare standard? Not necessarily. Their focus on limiting economic power, securing a contestable and rivalrous market structure, and fairness in digital markets does not mean that consumer interests are irrelevant for the new platform regulations. On the contrary, the new platform regulations rest on the assumption that greater structural rivalry and equality of opportunity at the level of *inter*-platform and *intra*-platform competition will ultimately result in greater innovation and consumer welfare in the medium-to-long run. They thereby operationalize the Arrovian assumption that greater rivalry will foster innovation efforts.⁶³

The new platform regulations thus stand for the proposition that more rivalry in digital markets, fostered by greater contestability and equality of opportunity, will benefit consumers in the medium-tolong run in the form of enhanced innovation, service quality, choice, and privacy.⁶⁴ Implicit in this proposition is the assumption that the interests of smaller innovative competitors, business users, and end users are largely aligned. It also implies that increased contestability and rivalry might enable or compel firms to differentiate themselves from incumbent platforms, for instance, by providing less privacy-intrusive products and services. Moreover, greater contestability and equality of opportunity among rivals are also thought to lead to lower prices for consumers because business users will be able to operate on lower costs and will pass on these lower costs to end users.

The new platform regulations thus embrace a more open, holistic and, arguably, more accurate understanding of consumer welfare that comprises alongside lower prices also non-price parameters such as quality, innovation, choice, and privacy. This holistic understanding of consumer welfare pays heed to the fact that many digital markets are zero-price markets on which firms compete for customer goodwill on the basis of non-price parameters. Most importantly, this holistic understanding of consumer welfare is cognizant of the fact that consumers also value things that are difficult to quantify in

This view has been notably coined by Philip Areeda & Donald F. Turner, Predatory Pricing and Related Practices under Section 2 of the Sherman Act, 88 Harv. L. Rev. 697 (1975) 709–710.

Brooke Group v. Brown & Williamson Tobacco Corp. supra note 58, at 224; Brown Shoe Co. Inc. v. United States 370 U.S. 294 (1962) 320; Case T-399/16 CK Telecoms UK Investments v Commission ECLI:EU:T:2020:217 para. 362.

^{62.} CARL KAYSEN & DONALD F. TURNER, ANTITRUST POLICY: AN ECONOMIC AND LEGAL ANALYSIS 7 (Harvard University Press 1959).

Kenneth J. Arrow, Economic Welfare and the Allocation of Resources for Invention, in THE RATE AND DIRECTION OF INVENTIVE ACTIVITY: ECONOMIC AND SOCIAL FACTORS (National Bureau Committee for Economic Research ed., 1962) 619–622, https:// www.nber.org/system/files/chapters/c2144/c2144.pdf.

^{64.} DMA Proposal, *supra* note 2, recitals 4, 7, 61, and 79; Explanatory Memorandum, at 1, 5, 10; Advice of the Digital Markets Taskforce, *supra* note 3, paras. 4; 2.8; DCMS/BEIS Consultation Document, *supra* note 3, paras. 28–30; Parliamentary Resolution on the Competition Law 4.0, *supra* note 23, at 9; Legislative proposal of the Competition Law 4.0, *supra* note 57, at 58, 71, 76.

monetary terms, such as privacy or the option to flexibly change their preferences in the future should the so far preferred choice prove no longer attractive or "turn rogue."⁶⁵

C. The Implications of the Multi-Value Approach: Managing Value Conflicts

Despite the continuous importance of consumer interests, the new platform regulations fundamentally diverge from the consumer welfare standard of conventional antitrust in so far as consumer interests are no longer considered the exclusive or primary goal that, in case of conflict, trumps all other considerations. This raises the central question of whether the various goals pursued by the new platform regulations always prove complementary to the extent their architects seem to assume. This is anything but certain. The history of the U.S. Robinson-Patman Act,⁶⁶ for instance, constitutes a prominent example of how antitrust statutes which pursue multiple, at times conflicting, objectives may easily run into difficulties and turn into an anti-consumer welfare policy.⁶⁷ The claim that competition law when pursuing a "hotchpotch" of various goals inevitably results in inconsistent outcomes was a central element of the Chicago School critique.⁶⁸ It also explains the breath-taking success of the consumer welfare standard which allowed modern antitrust to avoid goal conflicts by reducing the normative content of competition law to the single goal of consumer welfare.⁶⁹

The prospect of potential conflicts between the multiple policy goals pursued by the platform regulations invite three reflections. First, more empirical research is necessary to understand under which circumstances specific goals of the new platform regulations cease to be complementary and start pulling in opposite directions. Narrowing down these instances might enable legislators and enforcers to reduce goal conflicts to a minimum. Second, mechanisms should be devised that allow for a transparent prioritization and trading-off of various objectives in those remaining, hopefully rare but inevitable, occasions of goal conflicts. Third, instead of seeking to fully eradicate value conflicts, legislators and enforcers should endorse them pragmatically. Such value conflicts are, in fact, inevitable and coexistent with the indeterminacy of legal rules.⁷⁰ Attempts to define conflicts away by proclaiming that competition law should only pursue consumer welfare are therefore intellectually unsatisfactory and doomed to fail. They are also misleading because the adoption of the consumer welfare standard by modern antitrust constitutes nothing else than a political choice that seeks to settle value conflicts by declaring certain considerations irrelevant for deciding antitrust cases. Instead of sweeping them under the rug, legislators and enforcers should candidly address those value conflicts and clearly articulate and justify the political choices they make to settle them. As long as these choices are the results of a good-faith attempt to trace the interests of all relevant stakeholders and remain contestable through legal and political channels, they should be considered legitimate both in legal and democratic terms.

IV. The Recalibration of the Error-Cost Framework

A second significant reconfiguration brought about by the new platform regulations is the recalibration of the baseline assumptions of modern antitrust regarding the costs and benefits of competition

Burton A. Weisbrod, Collective-Consumption Services of Individual-Consumption Goods, 78 Q. J. ECON. 471 (1964); Tjalling C. Koopmans, On Flexibility of Future Preference, Cowles Foundation Discussion Paper 150 (1962), https://elischolar.library.yale.edu/cowles-discussion-paper-series/379/; David M. Kreps, A Representation Theorem for "Preference for Flexibility", 47 ECONOMETRICA 565 (1979).

^{66.} Robinson-Patman Act 1936. 15 U.S.C. § 13 et seq.

^{67.} For an example of such an inconsistent outcome, see Utah Pie Co. v. Continental Baking Co. 386 U.S. 685 (1967).

^{68.} For such a critique, see Robert H. BORK, THE ANTITRUST PARADOX: A POLICY AT WAR WITH ITSELF (Basic Books 1978) (1993).

^{69.} For an insightful discussion, see Herbert Hovenkamp, Antitrust Policy after Chicago, 84 MICH. L. REV. 213 (1985).

^{70.} Stavros Makris, Openness and Integrity in Antitrust, 17 J. COMPETITION LAW. ECON. 1, 35–36 (2021).

law intervention. Antitrust orthodoxy for a long time adhered to the view, initially coined by Chicago scholars,⁷¹ that as a matter of principle erroneous antitrust intervention tends to be more costly than erroneous non-intervention.

A. The Axioms of the Conventional Error-Cost Framework

This skewed understanding of error costs of modern antitrust rests on two axioms. First, it assumes that the probability of specific business conduct resulting in anticompetitive effects tends to be low.⁷² In other words, the orthodox error-cost framework is predicated on the belief that the prior probability of firm conduct being beneficial (P(B)) always exceeds the prior probability of it being harmful (P(H)), that is,

$$P(B) > P(H).$$

This in turn means that the odds ratio of the prior probability of business conduct being beneficial (P(B)) relative to it being harmful (P(H)) is always greater than 1,

$$\frac{P(B)}{P(H)} > 1^{73} \text{ or conversely } \frac{P(H)}{P(B)} < 1.$$

Second, the conventional error-cost framework also turns on the proposition that judicial errors are rarely corrected. Chicago scholars highlighted that erroneous convictions (i.e., type 1 errors or false positives) prevent not only efficiency-enhancing conduct in a single wrongly decided case but also deter procompetitive conduct by other firms in the future. This "judicial deterrence effect" is hence thought to amplify the costs of type 1 errors across time. Type 2 errors resulting from erroneous acquittals or non-intervention, by contrast, are presumed to be easily corrected by market forces. For, at least in the long run, monopoly profits will attract new entry that will erode market power. The preference of type 2 over type 1 errors encoded in the conventional error-cost framework thus embodies the belief that markets are robust and tend to self-correct.⁷⁴

This postulate of the disparate weight of error costs not only implies that the probability of conduct being truly anticompetitive is much lower than it being procompetitive, but it also suggests that the costs of erroneous antitrust intervention (Loss) often outweigh the accuracy benefits of antitrust intervention resulting from correctly averted or remedied antitrust harm (Gain), that is,

The skewed orthodox error-cost framework thus assumes that the ratio between the social losses created by the costs of false positives and the benefits resulting from accurate antitrust intervention that successfully averts or remedies truly harmful conduct always exceeds 1, that is,

$$\frac{Loss}{Gain} > 1 \text{ or conversely } \frac{Gain}{Loss} < 1.^{75}$$

75. Kaplow, supra note 73, at 17.

Frank H. Easterbrook, *The limits of Antitrust*, 63 TEX. L. REV. 1, 3, 15–16 (1984). The error-cost analysis has been trailblazed by Richard A. Posner, *An Economic Approach to Legal Procedure and Judicial Administration*, 2 J. LEG. STUD. 399 (1973). Isaac Ehrlich & Richard A. Posner, *An Economic Analysis of Legal Rulemaking*, 3 J. LEG. STUD. 257 (1974).

^{72.} See for instance for this proposition BORK, *supra* note 68, at 148–154 (predatory pricing), 303–307 (exclusive dealing), 380–381 (tying).

^{73.} The following formal analysis draws on Louis Kaplow, *Likelihood Ratio Tests and Legal Decision Rules*, 16 Am. LAW ECON. REV. 1, 11 (2014).

BORK, *supra* note 68, at 143–144; Easterbrook, *supra* note 71, at 15; Alan Devlin & Michael Jacobs, *Antitrust Error*, 52
 WM. & MARY L. REV. 75, 84–85 (2010).

It directly follows from this slanted understanding of error costs that under-enforcement (that means excess type 2 errors) constitutes the optimal standard of intervention or deterrence. This is because type 1 errors are more likely (as the likelihood of pro-competitive conduct exceeds that of anticompetitive conduct (i.e., [P(B) / P(H)] > 1) and create higher costs than type 2 errors (i.e., [Loss / Gain] > 1).

B. From Under- to Over-Enforcement as the Optimal Policy Standard

The orthodox concern that over-enforcement would unduly deter pro-competitive firm conduct for a long time dominated the academic and policy debate about the appropriate role of antitrust law, notably in innovation-driven digital markets. Market power or large market shares, it was often argued, were short-lived in digital markets as incumbents would remain constrained by dynamic, potential competition "for the market."⁷⁶ The importance of innovation and dynamic efficiencies in the digital sector further added to the costs the conventional error-cost framework associates with "judicial over-deterrence." It was feared that type 1 errors would not only deter other firms from engaging in conduct that maximizes static efficiencies but also dampen their incentives to innovate and generate dynamic efficiencies. Antitrust scholars and courts relied on a Schumpeterian⁷⁷ understanding of innovation, which posits that the prospect of temporary monopoly profits operates as an important incentive for firms to innovate,⁷⁸ to support the claim that heavy-handed competition law intervention against large incumbents in innovation-driven markets would inevitably harm innovation.⁷⁹

The new platform regulations fundamentally challenge this orthodox understanding of error costs and, with it, the widely held opposition to interventionist antitrust policy in technology-enabled markets. By subjecting digital platforms to ex ante rules and expanding the intervention toolkit of competition enforcers in digital markets, the new platform regulations instead stand for the proposition that the welfare costs of under-enforcement of traditional competition rules (type 2 errors) in digital markets outweigh the potential error costs caused by the over-enforcement of more or less broadly construed ex ante rules (type 1 errors). The new platform regulations thus openly challenge the assumption of orthodox antitrust that under-enforcement is the optimal standard of intervention in innovation-driven markets. At the same time, they also call into question the "contestable market hypothesis"⁸⁰ which posits that market power in digital markets is short-lived, as incumbents remain constrained by dynamic, potential competition "for the market." The new platform regulations, instead, suggest that in the case of doubt competition law enforcement in digital markets should err on the side of intervention and type 1 errors. The platform regulations hence embody the proposition that over-enforcement constitutes the optimal intervention standard for competition law in digital markets.

C. The Reasons for a Recalibrated Understanding of Error Costs

What explains this recalibration of the error-cost framework? The departure from the orthodox errorcost framework is representative of a broader reckoning that its preference for type 2 errors is grounded in very strong assumptions about the self-healing forces of markets and the costs of state

See for this view, Miguel Rato & Nicolas Petit, *Abuse of Dominance in Technology-Enabled Markets: Established Standards Reconsidered*?, 9 Eur. COMPETITION J. 1, 9–10 (2013); Case T-79/12 Cisco v Commission ECLI:EU:T:2013:635 para. 69.

^{77.} JOSEPH A. SCHUMPETER, CAPITALISM, SOCIALISM AND DEMOCRACY Chapter VIII (Harper & Brothers, 1942) (1962).

See for a recent restatement of this position FTC v. Qualcomm Inc. 969 F. 3d 974 (2020) 990. Opinion of Advocate General Wahl in Case C-177/16 Autortiesību un komunicēšanās konsultāciju aģentūra/ Latvijas Autoru apvienība v Konkurences padome ECLI:EU:C:2017:286 para. 117.

^{79.} See for instance Rato Petit, *supra* note 76, at 9–10; John G. Sidak and David J. Teece, 'Dynamic Competition in Antitrust Law' 5 J. COMPETITION LAW. ECON. 581, 611–612 (2009); Nicolas Petit & David J Teece, *Innovating Big Tech Firms and Competition Policy: Favoring Dynamic over Static Competition*, 30 IND. CORP. CHANG. 1168, 1170, 1185, 1188 (2021).

^{80.} WILLIAM J. BAUMOL ET AL., CONTESTABLE MARKETS AND THE THEORY OF INDUSTRY STRUCTURE (Harcourt College 1982).

intervention.⁸¹ There is hence a growing awareness that the orthodox error-cost framework is all too often rooted in inherently political value judgments and beliefs—namely, a strong ideological aversion against state intervention and preference for *laissez-faire*, *laissez-aller*—rather than in market realities.

Post-Chicago scholarship has notably shown that the Chicago School analysis on which the orthodox error-cost framework is premised underestimated the frequency at which unilateral conduct by powerful firms may result in anticompetitive harm.⁸² This reconsideration of the probability distribution of anticompetitive conduct has been further compounded by the specific features of digital markets. The presence of extreme economies of scale and scope, network effects, and rising levels of industry concentration is believed to make digital markets more prone to anticompetitive outcomes than other industries.⁸³ Recent economic scholarship thus questions the fundamental premise of the orthodox error-cost framework that firm conduct always tends to be more likely to be beneficial than harmful. Instead of assuming that the prior odds ratio of pro-competitive effects always exceeds 1 ([P(B)/P(H)] > 1), there is a heightened concern that specific types of platform conduct are more likely to result in anticompetitive harm than benefits. Accordingly, there is a growing recognition that the ratio of prior probabilities of beneficial versus harmful effects associated with specific platform conduct may fall short of 1, that is,

$$\frac{P(B)}{P(H)} < 1$$
 or conversely $\frac{P(H)}{P(B)} > 1$.

The recalibration of the error-cost framework underpinning the new platform regulation is, however, not only informed by the recognition that anticompetitive outcomes occur in digital markets more frequently than it was believed by the orthodox error-cost framework. The revisited assessment of error costs also gives currency to the view that anticompetitive conduct by powerful digital platforms tends to result in harm of substantial magnitude.⁸⁴ First, owing to their size and importance as gateways to online markets and customer groups, anticompetitive conduct by powerful platforms is likely to affect a vast number of transactions by end and business users. Second, the propensity of digital markets to tip on a lasting basis in favor of the predominant incumbent further amplifies the harm of anticompetitive conduct by rendering it irreversible (or at least prohibitively costly to reverse).⁸⁵ As network effects might tilt digital markets on a lasting basis, market mechanisms are less likely to readily self-correct any type 2 errors than the orthodox error-cost framework assumes. Given the low degree of contestability, once a digital market has pivoted in favor of a powerful incumbent, competing alternative innovation paths and technological solutions may be forever lost.

The magnitude of harm of anticompetitive outcomes in digital markets has, therefore, an important inter-temporal dimension. If markets by reason of the presence of strong network effects and limited contestability are unlikely to rapidly eradicate anticompetitive outcomes, the magnitude of anticompetitive harm increases the longer it takes competition authorities or courts to intervene. The magnitude of anticompetitive harm, in other words, is positively correlated with the size and importance of digital platforms as gateways and negatively correlated with market contestability, reversibility of anticompetitive harm, and the speed of competition law intervention.

This growing awareness of the significant scale of the costs of false negatives inverts the ratio between the losses and gains of antitrust intervention. The revised error-cost framework underpinning

^{81.} Jonathan B. Baker, Taking the Error out of "Error Cost" Analysis, 80 ANTITRUST LAW J. 1 (2015).

Louis Kaplow, Extension of Monopoly Power through Leverage, 85 COLUMBIA LAW REV. 515 (1985). Jonathan B. Baker, Recent Developments in Economics that Challenge Chicago School Views, 58 ANTITRUST LAW J. 645 (1989); Krattenmaker & Salop, supra note 44; Michael D. Whinston, Tying, Foreclosure, and Exclusion, 80 AM. ECON. Rev. 837 (1990).

^{83.} Furman et al., *supra* note 39, 1.65–1.92. Crémer et al., *supra* note 39, at 4, 15–16, 50.

^{84.} Furman et al., supra note 39, paras. 3.88–3.100; Crémer et al., supra note 39, at 50–51.

On the concept of irreversibility, Kenneth J. Arrow & Anthony C. Fisher, *Environmental Preservation, Uncertainty, and Irreversibility*, 88 Q. J. ECON. 312 (1974).

the new platform regulations posits that, at least in digital markets, the gains of antitrust intervention often outweigh its losses. Instead of assuming that the ratio between losses and gains of antitrust intervention always exceeds 1 ($\frac{Loss}{Gain} > 1$), the new platform regulations are predicated on the assumption

that the ratio may in certain circumstances fall below 1, that is,

$$\frac{Loss}{Gain} < 1$$
, or conversely $\frac{Gain}{Loss} > 1$.

The revised error-cost framework underpinning the new platform regulations thus epitomizes not only a reconsideration of the probability distribution of anti- and procompetitive effects but also evinces greater awareness for the magnitude of harm that anticompetitive conduct may bring about in digital markets. Anticompetitive harm in digital markets tends to be of particularly sizeable scale if antitrust intervention and market mechanisms are slow in correcting anticompetitive outcomes and/or innovation is irreversibly lost as a consequence of it. The orthodox error-cost framework was grounded in the apprehension that judicial errors will stay uncorrected and procompetitive conduct will be forever deterred and lost. The recalibrated error-cost framework of the new platform regulations turns this assumption upside down. It recognizes that by reason of network effects and the tendency of markets to tip on a lasting basis in favor of the incumbent, anticompetitive harm may stay forever uncorrected. In the presence of a higher frequency of anticompetitive and potentially irreversible harm of substantial magnitude, over- rather than under-deterrence becomes the optimal standard of intervention. Instead of preferring type 2 errors, it suddenly makes economic sense to err on the side of type 1 errors.⁸⁶

V. A Reconfiguration of the Modus Operandi of Modern Antitrust

The recalibration of the error-cost framework envisaged by the new platform regulations finds its direct and most significant expression in a third major shift, namely the reconfiguration of the modus operandi of modern antitrust. The new platform regulations curtail the extent to which antitrust enforcers are required to rely on case-specific information or evidence to make a legal determination of the (il)legality of specific conduct. All new platform regulations thereby fundamentally depart from the incremental shift of modern antitrust toward a more inductive competition law analysis. Instead, they resurrect a more deductive mode of decision making that relies on strong priors to infer anticompetitive harm from the economic and legal form of specific platform conduct. To illustrate this transformation, this section proposes a basic model of antitrust decision making (A) before describing how the new platform regulations will alter this mode of decision making (B) and considering the implications of this development (C).

A. A Basic Model of Antitrust Decision Making

The decision making of competition authorities and judges in antitrust cases can be modeled by Bayesian decision theory. In competition proceedings, the basic task of the factfinder consists of deciding whether impugned conduct will create anticompetitive harm (H) or will be competitively neutral/beneficial (B or non-harm). Bayes' theorem, which forms the basis of subjectivist approaches to epistemology and theories of evidence,⁸⁷ describes how such a determination of the anticompetitive nature of specific impugned conduct can be formed. This decision-making process can be expressed in the following formula:

$$\frac{P(H)}{P(B)} \times \frac{P(e|H)}{P(e|B)} = \frac{P(H|e)}{P(B|e)}$$

^{86.} Id. at 318-319.

^{87.} James Joyce, Bayes' Theorem (2003), https://plato.stanford.edu/entries/bayes-theorem/.

Bayes' theorem outlines how a factfinder can form probability estimates by updating prior beliefs with further evidence. To this end, the factfinder will build a hypothesis about the impugned conduct's impact on competition by first forming a prior probability opinion. This prior probability estimate [P(H) / P(B)] offers a first prediction (or guess) about the conduct's likelihood to result in anti-competitive harm (P(H)) relative to its probability to result in non-harm (P(B)). The odds ratio [P(H) / P(B)] of this prior probability estimate is derived from easily observable facts or beliefs about the (anti-)competitive nature or incidence (i.e., distribution of harmful and beneficial effects)⁸⁸ of specific forms of business conduct.

The prior probability estimate [P(H) / P(B)] can subsequently be revised and updated in the light of additional case-specific evidence *e* about the impact of the impugned conduct on competition in the specific case at hand [P(e|H) / P(e|B)]. The information *e* is gathered on a case-by-case basis during the competition proceedings. The likelihood ratio (LR) [P(e|H) / P(e|B)] describes the conditional probability of finding specific evidence *e* (e.g., high market shares, price or output effects, foreclosure effects, internal documentation) given that the impugned conduct is truly harmful (*e*|*H*) relative to the conditional probability of the evidence at hand being associated with a truly efficient, beneficial, or competitively neutral conduct (*e*|*B*).

By combining the prior probability [P(H)/P(B)] and the likelihood ratio of additional evidence [P(e|H)/P(e|B)], the factfinder will then establish a revised estimate of the conduct's posterior probability [P(H|e)/P(B|e)] to result in anticompetitive harm (compared with the probability that it is beneficial). At the end of the competition law proceeding, the factfinder will have to make a determination about the competitive nature of the impugned conduct based on this posterior probability estimate that the conduct is harmful (relative to that it is not) given the additional evidence *e*.

The above expression of the Bayesian theorem hints at two alternative, albeit complementary, modes of decision making competition authorities and courts can use to make determinations about the lawfulness of specific conduct.⁸⁹ The first mode of decision making is deductive in nature. Under this deductive mode, a factfinder can primarily rely on prior probability estimates of the impugned conduct entailing anticompetitive harm. This prior can take the form of legal rules or presumptions that indicate the legality of specific types of conduct without requiring the factfinder to account for a broad range of additional assessment criteria. Rules or presumptions operate as analytical shortcuts or heuristic devices which enable a factfinder to infer fact B from the showing of another fact A.⁹⁰ They thus allow a factfinder to infer a certain legal fact or conclusion, such as the anticompetitive nature or illegality of specific evidence.⁹¹ Instead of proving the anticompetitive nature of the impugned conduct on the basis of case-specific evidence showing that it is actually or likely causing anticompetitive harm, a competition authority or court can deduce the anticompetitive harm from the way in which its legal or economic form relates to a specific rule or presumption.

The respective weight of this presumption (prior) and additional evidence e can vary. The presumption can be neutral (i.e., equal 1) if the competition authority starts from the premise that the

Arndt Christiansen & Wolfgang Kerber, Competition Policy with Optimally Differentiated Rules Instead of per se Rules vs. Rule of Reason, 2 J. COMPETITION LAW ECON. 215, 229 (2006).

Mark A. Lemley & Christopher R. Leslie, *Categorical Analysis in Antitrust Jurisprudence*, 93 Iowa Law Rev. 1207, 1212– 1219 (2007).

Steven C. Salop, An Enquiry Meet for the Case: Decision Theory, Presumptions, and Evidentiary Burdens in Formulating Antitrust Legal Standards 2 (2017) https://scholarship.law.georgetown.edu/facpub/2007/; Andriani Kalintiri, Analytical Shortcuts in EU Competition Enforcement: Proxies, Premises, and Presumptions, 16 J. COMPETITION LAW ECON. 392, 397 (2020).

^{91.} For recent discussions of different types of presumptions and their role in U.S. antitrust and E.U. competition law, see Andrew I. Gavil & Steven C. Salop, *Probability, Presumptions and Evidentiary Burdens in Antitrust Analysis: Revitalizing the Rule of Reason for Exclusionary Conduct*, 168 UNIV. PA LAW REV. 1 (2020). ANDRIANI KALINTIRI, EVIDENCE STANDARDS IN EU COMPETITION ENFORCEMENT: THE EU APPROACH 142–168 (Bloomsbury Publishing 2019).

respective probabilities of the conduct being anticompetitive or not are on par with each other ([P(H) / P(B)] = [0.5 / 0.5] = 1). Depending on the direction in which it pulls ([P(H) / P(B)] > < 1), the presumption tilts the balance toward finding anticompetitive harm (*H*) or not (*B*).

The strength of a legal presumption thus regulates the amount of information a competition authority is required to account for and produce in order to make a legal qualification of the impugned conduct. It operates as a multiplier that affects the weight of additional evidence or information a factfinder needs to consider to assess the legality of the conduct. At the same time, it also determines the height of the evidentiary burden a defendant has to meet to rebut a presumption of anticompetitiveness operating against its impugned conduct.⁹² The higher (lower) the prior odds ratio, the stronger the additional evidence must be to offset (confirm) the initial presumption of anticompetitive harm. A legal presumption thus puts the "thumb on the scale" as it tilts the decision making toward the finding of anticompetitive harm (or non-harm).⁹³

The second mode of antitrust decision making is inductive in nature. It consists of a careful and casuistic inquiry into the actual or likely effects of the impugned conduct on competition. This inductive mode of decision making fundamentally differs from a deductive analysis where the prior carries significant weight as the factfinder relies on hard-and-fast rules or strong presumptions and infers the legality of specific conduct from its economic and legal form. The inductive method, instead, requires the factfinder to consider additional assessment criteria and case-specific evidence to form an opinion as to whether the conduct has actual or likely anticompetitive effects. The importance of this case-specific evidence is negatively correlated to the strength of the prior. The weaker the prior, the more case-specific information the factfinder starts from the prior that the conduct is competitively neutral ([P(H)/P(B)] = [0.5/0.5] = 1), the determination of the anticompetitive nature of conduct depends exclusively on the assessment of case-specific evidence.

Most of the time, antitrust decision making will not take the pure form of either deductive or inductive analysis. Rather, both modes describe ideal types that lie at the two extremes of a continuum of more or less differentiated modes of analysis and rules.⁹⁴ Real-life decision making will often be a mix of and thus sit somewhere in between these two extremes. Whether decision making gravitates more to the deductive or the inductive approach depends on the exact weight of the prior and the case-specific evidence in the formation of the posterior probability estimate.

Nonetheless, historically competition law analysis on both sides of the Atlantic increasingly drifted under the banner of the "more economic" or "effects-based" approach toward a more inductive mode of decision making.⁹⁵ The primary channel through which this shift toward the inductive, effects-based approach took shape was the reconfiguration of the type of "legal commands"⁹⁶ competition law chooses to lay down proscriptions and obligations; in short, the manner in which competition law formulates propositions of normative desiderata.⁹⁷ With the rise of the so-called "more economic approach" under the auspices of the Chicago School, flexible legal standards that call for a casuistic and inductive analysis of "demonstrable economic effects"⁹⁸ became the preferred default mode of modern antitrust analysis. The reliance of U.S. and E.U. competition law on self-executing rules or broadly construed

^{92.} Salop, *supra* note 90, at 26–29.

^{93.} Id. at 3.

^{94.} Christiansen & Kerber, supra note 88, at 220-225.

^{95.} For a comprehensive analysis, ANNE WITT, THE MORE ECONOMIC APPROACH TO EU ANTITRUST LAW (Bloomsbury Publishing 2016).

^{96.} Louis Kaplow, Rules versus Standards: An Economic Analysis, 42 Duke Law J. 557, 559-561 (1992).

^{97.} For the notion of rules as normative desiderata, Christian List, *Republican Freedom and the Rule of Law*, 5 POLIT. PHILOS. ECON. 201, 205 (2006).

^{98.} Cont'l T.V. v. GTE Sylvania 433 U.S. 36 (1978) 58-59; Leegin Creative Leather Prods. v. PSKS, Inc. 551 U.S. 877 (2007) 887.

legal presumptions was increasingly disavowed as outdated "formalistic line drawing"⁹⁹ that led to economically illiterate outcomes. Modern antitrust thus increasingly opted for more differentiated and complex legal commands, taking the form of flexible standards that require competition authorities to account for additional, case-specific assessment criteria (i.e., greater amounts of evidence e) to enhance the precision of antitrust analysis in telling apart truly harmful from truly beneficial conduct.¹⁰⁰

Modern antitrust's preference for an inductive method relying on "flexible" standards rather than "rigid" rules was the immediate result of the rise of the orthodox error-cost framework and its aversion against type 1 errors. As long as the axiom of the orthodox error-cost framework that, as a matter of principle, business conduct is most of the time pro-competitive—that is, [P(B)/P(H)] > 1 or conversely [P(H)/P(B)] < 1—holds true, deductive antitrust decision making relying on strong priors (i.e., [P(H)/P(B)] > 1) in the form of presumptions of illegality or per serules turns out to be by its very nature over-inclusive. The frequent reliance of antitrust law preceding the more economic approach on rule-like presumptions was therefore perceived as a major source of too many type 1 errors. To align antitrust decision making with the axiom of the orthodox error-cost framework (i.e. [P(H)/P(B)] < 1) and reduce type 1 errors, modern competition law opted for greater differentiation and more complex legal commands.¹⁰¹ While turning down the weight of priors (i.e., [P(H) / P(B)]), it increased the weight of the assessment of additional case-specific evidence (i.e., [P(e|H)/P(e|B)]) in the overall determination of the antitrust legality of specific conduct: $\downarrow [P(H) / P(B)] \times \uparrow [P(e|H) / P(e|B)] = [P(H|e) / P(B|e)]$. The endorsement of the orthodox errorcost framework under the auspices of the Chicago School and the more economic approach in Europe thus translated itself into a recalibration of the mode of decision making and differentiation of legal commands that reduced the role of deductive to the benefit of inductive reasoning.

B. From Standards to Rule-Like Presumptions

The new platform regulations mark a significant rupture in what appeared an inexorable progress toward a more inductive, effects-based approach and a greater differentiation of legal commands in competition law. Some of the new platform regulations rely on rebuttable¹⁰² presumptions to identify firms that qualify as designated platforms.¹⁰³ All new platform regulations also introduce novel self-executing, rule-like presumptions of anticompetitiveness for specific forms of unilateral platform conduct¹⁰⁴ or mergers¹⁰⁵ (see Tables 2 and 3). They thus establish ex ante bans preventing designated platforms from engaging in different forms of "blacklisted" conduct that are deemed to be "particularly unfair or harmful" to competitors and/or reduce the contestability of markets.¹⁰⁶ Most of these presumptions codify a number of theories of harm that lay at the core of recent antitrust cases and investigations against powerful online platforms (see Table 5).

^{99.} Cont'l T.V. v. GTE Sylvania, supra note 98, at 58–59. Leegin Creative Leather Prods. v. PSKS, Inc. supra note 98, at 887. For a similar criticism in the E.U. context, Opinion of Advocate General Wahl in Case C-413/14 P Intel v Commission ECLI:EU:C:2016:788 paras. 59–106.

^{100.} Christiansen & Kerber, supra note 88, at 217-19.

^{101.} id.

^{102.} DMA Proposal, supra note 2, art. 3 (4) and (6) and recital 23.

^{103.} Id. art. 3 (2) and (3). PCO Act, supra note 4, s. 3 (d); ACIO Act, supra note 4, s. 2 (d) and (g) (4). Schnitzer et al., supra note 5, at 9.

^{104.} DMA Proposal, *supra* note 2, arts. 5 and 6; Advice of the Digital Markets Taskforce, *supra* note 3, paras. 4.33–4.44, 4.50; DCMS/BEIS Consultation Document, *supra* note 3, para. 90; Competition Law 4.0, *supra* note 1, s. 19a (2); ACIO Act, *supra* note 4, s. 2 (a) and (b); ACCESS Act, *supra* note 4, s. 3 and 4. For a comprehensive analysis, see Anne C. Witt, *Platform Regulation in Europe—per se Rules to the Rescue?*, 18 J COMPETITION LAW ECON. 13–37 (2022).

^{105.} PCO Act, *supra* note 4, s. 2 (a).

^{106.} DMA Proposal, supra note 2, recitals 32-58; arts. 5 and 6.

| Prohibitions/Obligations | Relevant Cases or Enforcement Activity |
|---|--|
| Cross-usage of data collected through different proprietary/third party services | Autorità Garante della Concorrenza et del Mercato (AGCM), ENEL ¹⁰⁷ / Case C-377/20 Servizio Elettrico Nazionale and Others ¹⁰⁸ Autorité de la Concurrence (AdC) Décision, 17-D-06 du 21 mars 2017 relative à des pratiques mises en œuvre dans le secteur de la fourniture de gaz naturel, d'"électricité et de services énergétiques (GDF Suez) ¹⁰⁹ AdC, Décision n° 09-D-24 du 28 juillet 2009 (France Télécom) ¹¹⁰ AdC, Avis n°10-A-13 du 14 juin 2010 relatif à l'utilisation croisée des bases de clientele ¹¹¹ Belgian Competition Authority, Decision 2015-P/K-27 of 22 September 2015 Stanleybet Belgium/Stanley International Betting and Sagevas/ World Football Association/Samenwerkende Nevenmaatschappij Belgische PMU v. Nationale Lotenj ¹¹² |
| Use of non-public data sourced from other business users active on the platform | European Commission, Amazon investigation ¹¹⁴ European Commission, Apple investigation ¹¹⁵ CMA, Facebook investigation ¹¹⁶ |
| Data portability Acress to data anorated by husiness users | Contractual restrictions on (campaign) data portability on Google's AdWords service ¹¹⁷ European Commission Apole investigation ¹¹⁸ |
| Data sharing with competitors | european Commission, Apple investigation AdC, Décision 17-D-06 du 21 mars 2017 relative à des pratiques mises en œuvre dans le secteur de la fourniture de gaz naturel, d"électricité et de services énergétiques (GDF Suez) |
| Rebalancing of asymmetric bargaining power | Bundeskartellamt, Facebook ¹¹⁹ Bundeskartellamt, A <i>mazon</i> ¹²⁰ CMA, Google's privacy sand box investigation ¹²¹ CMA, Market study of online advertising markets ¹²² |
| Interoperability | European Commission, Microsoft ¹²³ AGCM. Google/ENEL ¹²⁴ AGCM. Google (Display advertising) ¹²⁵ |
| Self-preferencing | European Commission, <i>Google Shopping</i> ¹²⁶ European Commission, Amazon investigation ¹²⁷ CMA, Google's privacy sandbox investigation ¹²⁸ CMA, Facebook investigation ¹²⁹ AdC, <i>Google¹³⁰</i> AGCM Amazon ¹³¹ |
| ΩFN | E.U. and U.S. Apple e-book cases Erforcement. S. Apple e-book cases Erforcements activities of various competition authorities against MFNs in the hotel booking platform sector ¹³² |
| Anti-steering | Epic Games v Apple ¹³³ European Commission, Apple investigations ¹³⁴ CMA, Apple App store investigation ¹³⁵ |
| Tying and defaults | European and U.S. <i>Microsoft</i> cases ¹³⁶ European Commission, <i>Microsoft tying¹³⁷</i> European and U.S. <i>Google Android</i> ¹³⁸ European Commission, Apple Pay investigation ¹³⁹ European Commission, Apple investigation ¹³⁹ |
| (Secondary-line) discrimination | European Commission, Apple Pay investigation ¹⁴¹ |

| | Provvedimento n. 27494 A511—ENEL/CONDOTTE ANTICONCORRENZIALI NEL MERCATO DELLA VENDITA DI ENERGIA ELETTRICA 2018. Opinion of Advocate General Rantos in Case C-377/20 Servizio Elettrico Nazionale and Others ECLI:EU:C:2021:998. |
|-----|--|
| 326 | 109. Décision 17-D-06 du 21 mars 2017 relative à des pratiques mises en œuvre dans le secteur de la fourniture de gaz naturel, d'électricité et de services énergétiques (GDF Suez) 2017. 110. Décision n° 09-D-24 du 28 juillet 2009 relative à des pratiques mises en oeuvre par France Télécom sur différents marchés de services de communications électroniques fixes dans les DOM 2009, Décision n° 09-D-24 du 28 juillet 2009 (France Télécom). |
| | Avis n°10-A-13 du 14 juin 2010 relatif à l'utilisation croisée des bases de clientèle 2010. Decision 2015-P/K-27 of 22 September 2015 Stanleybet Belgium/Stanley International Betting and Sagevas/World Football Association/Samenwerkende Nevenmaatschappij Belgische PMU v. Nationale Loterii. 2015. |
| | 113. Bundeskartellamt, Facebook Inc. i.a, B6-22/16. 114. Furnean Commission. Press Release 1P/20/2077—Antitrust: Commission Sends Statement of Objections to Amazon for the Use of Non-Public Independent Seller. Data and Opens Second |
| | |
| | 115. Case No COMPTATAUTIO Applie—App Store Fractices. 116. Competition and Markets Authority, <i>CMA Investigates Facebook's Use of Ad Data</i> (2021), https://www.gov.uk/government/news/cma-investigates-facebook-s-use-of-ad-data. |
| 1 | 117. Google Inc, Commitments Letter From Google Inc. To Chairman Leibowitz, Google Inc. FTC-File No. 111-0163 (Dec. 27, 2013), https://www.ftc.gov/system/files/documents/closing_letters/ |
| 1 | googte-inc./130105googteteterenairmanterbowrtz.put. 118. Case No COMP/AT.40716 Apple—App Store Practices <i>supra</i> note 115. |
| 1 | 119. Bundeskartellamt, Facebook Inc. i.a, B6-22/16 supra note 113. |
| 1 | Bundeskartellamt, Online Sales (Amazon), B2-88/18. |
| _ | |
| _ | |
| _ | |
| | |
| _ | |
| - | |
| 1 | European Commission supra note 114. |
| - | |
| | |
| | |
| - | Autorità garante della Concorrenza e del Mercato, Press Release A528—Antitrust—Amazon Fined over € 1,128 Billion for Abusing Its Dominant Position (2021), https://en.agcm.it/en/media/ mese.releases/2021/12/4 528 |
| - | 13 Cons No. COMB/AT 1512 - Boold MDN and solved motions. C/0017) 2026 Each ILC 14 Analy Inc. 053 B Comm 24:239 /0013). Example in Manualy Inc. do Manufacian Manualy Inc. do Manualy Inc. 2020. Each 2020 - 202 |
| - | |
| | Broad Retail Price MFN Clauses: Are they RPM "At Its Worst?" 81 ANTITRUST LAW J. 65 (2016). |
| - | 133. Epic Games, Inc. v. Apple Inc. 20-cv-05640-YGR. |
| | 134. Case No COMP/AT.40716 Apple—App Store Practices supra note 115; Case No COMP/AT.40437 Apple—App Store Practices (music streaming); Case No COMP/AT.40652 Apple—App |
| | |
| - | 135. Competition and Markets Authority (CMA), CMA Investigates Apple over Suspected Anti-Competitive Behaviour, https://www.gov.uk/government/news/cma-investigates-apple |
| | |
| | |
| | |
| | 138. Case No COMP/AI.40099 Google Android. C(2018) 4/61 final; Complaint against Google filed in the United States District Court for the District of Columbia 20 October 2020. Case 1:20-cv-03010. |
| | |
| | Store Practices (e-books/audiobooks). |
| - | 141. Case No COMP/AT.40452 Apple—Mobile payments <i>supra</i> note 140. |

By codifying existing or novel theories of harm into rule-like presumptions, the new platform regulations assign increased weight to priors in the determination of anticompetitive conduct in digital markets. Consequently, they importantly depart from the preference of modern antitrust for flexible legal standards and an ever more inductive, casuistic analysis of anticompetitive effects. The new platform regulations bring about a partial roll-back of the inductive "effects-based" approach and more differentiated legal standards to the benefit of a more deductive "form-based" approach and less differentiated rules.¹⁴² They are thus suggestive of a transition from an equilibrium characterized by the decreasing weight of priors ($\downarrow [P(H) / (P(B)] \times \uparrow [P(e|H) / P(e|B)] = [P(H|e) / P(B|e)]$) to a new mode of decision making marked by the greater importance of priors relative to case-specific evidence ($\uparrow [P(H) / P(B)] \times \downarrow [P(e|H) / P(e|B)] = [P(H|e) / P(B|e)]$).

This greater role of rule-like presumptions not only reduces the amount of information competition authorities have to process before they can qualify platform conduct as unlawful, but it also substantially heightens the evidentiary burden for defendants. All new platform regulations indeed attribute considerable weight to the newly crafted presumptions by limiting the degree to which they can be rebutted by designated platforms. The presumptions established in arts. 5 and 6 of the DMA arguably carry the most important weight. Although they remain rebuttable, the DMA sets a particularly demanding standard for defendants to reverse them. The DMA does not recognize any form of efficiency defense that defendants could advance to justify blacklisted conduct. Rather, designated platforms can only defeat the presumptions by putting forth offsetting evidence showing that the conduct is exempted under public interest grounds¹⁴³ or temporarily necessary to secure the viability of their business.¹⁴⁴

The U.S. proposals also set a high evidentiary burden for the rebuttal of presumptions of illegality for unilateral platform conduct or platform acquisitions. Unlike conventional antitrust law, the new proposals do not provide for a clear-cut efficiency defense. Instead, designated platforms can merely rely on an "affirmative defense" to rebut presumptions of illegality.¹⁴⁵ Under the proposed rules for platform mergers, designated platforms can also obtain an "exemption" from the prohibition against platform acquisitions if they demonstrate that the proposed merger does not have any adverse effect on actual, potential, or nascent competition.¹⁴⁶ To rebut these presumptions, designated platforms must, however, meet an exacting evidentiary burden. They have to produce "clear and convincing evidence"¹⁴⁷ suggesting that the conduct or merger is with a strong probability—of 75 percent or more¹⁴⁸—competitively neutral or pro-competitive.

The presumptions listed in s. 19a GWB carry considerable weight, too. Yet, designated platforms appear to have to surmount a less important evidentiary hurdle to overturn them. Although s. 19a does not provide for any explicit efficiency defense or other public policy–related exemptions, the German legislator clearly envisaged the possibility for addressees of s. 19a to advance objective justifications in defense of the presumed abusive forms of conduct.¹⁴⁹ By contrast, the presumptions under the SMS

^{142.} For a discussion of the dichotomy between a "form-based" and "effects-based" approach, see, Patrick Rey & James S. Venit, *An Effects-Based Approach to Article 102: A Response to Wouter Wils*, 38 WORLD COMPETITION 3 (2015).

^{143.} DMA Proposal, *supra* note 2, art. 9.

^{144.} Id. art. 8.

^{145.} ACIO Act, supra note 4, s. 2 (c).

^{146.} PCO Act, *supra* note 4, s. 2 (b).

^{147.} ACIO Act, supra note 4, s. 2 (c); PCO Act, supra note 4, s. 2 (b).

^{148.} For this operationalization of the "clear and convincing evidence" standard, see Michelle Burtis, Jonah B. Gelbach, & Bruce H. Kobayashi *Error Costs, Legal Standards of Proof and Statistical Significance*, 25 SUPREME COURT ECON. REV. 1, 18 (2017). The clear and convincing evidence standard thus requires the designated platforms to proffer evidence showing that the impugned conduct or planned merger is three times more pro- than anticompetitive. Conversely, this suggests that the U.S. platform regulations rely on a prior that assumes that specific platform conduct or mergers are three times more likely to be anticompetitive than not.

^{149.} Competition Law 4.0, supra note 1, s. 19a (2) penultimate sentence. Parliamentary Resolution on the Competition Law 4.0, supra note 23, 113, 116–17; Franck & Peitz, supra note 10, at 521–22.

regime have a considerably less important weight because the SMS framework appears to accept a broader range of admissible rebuttal evidence. Unlike the DMA, the SMS framework recognizes the possibility for firms to plead an efficiency defense or objective justification to save conduct that otherwise violates their code of conduct.¹⁵⁰

In relying primarily on rule-like presumptions of considerable weight to lay down legal commands for digital platforms, the new platform regulations mark an inflection point in the trend from deductive to inductive decision making that shaped the more effects-based or economic approach and its operationalization through greater differentiation of legal commands. The immediate consequence of this reversal toward a more deductive mode of decision making and less differentiated rules is that competition authorities can more easily rely on prior probability estimates of anticompetitive harm to make a determination about the (un)lawful nature of the conduct at hand. The corollary of the revaluation of priors and decreased differentiation is the declining weight and amount of case-specific evidence that competition authorities are required to assess before impugned conduct can be legitimately qualified as anticompetitive.

C. Implications of the Increased Weight of Rule-Like Presumptions

This pivot toward a more deductive mode of decision making and the concomitant choice of priors, in the form of less differentiated rule-like presumptions, rather than differentiated, open-textured standards, to lay down legal commands and prescriptions for powerful platforms, has a number of significant implications.

The most important implication is that rule-like presumptions, unlike standards, have a modal character. They lay down moral desiderata that are considered true not only in the specific, actual world (that is, the world where a competition authority or court finds based on case-specific evidence that a firm has unduly interfered with another firm/competition) but across all relevant possible and legally permissible worlds.¹⁵¹ Legal rules and rule-like presumptions thus formulate normative propositions or legal commands whose realization and enforcement is less contingent on changing, case-specific circumstances than legal standards that inform the inductive mode of decision making of modern antitrust law. Owing to their modal character, the rule-like presumptions enshrined in the new platform regulations follow a clearly preventative character. They seek to prevent designated platforms from engaging in certain forms of conduct that have a high probability of leading to anticompetitive outcomes or may result in anticompetitive harm of significant magnitude because they further entrench the market power of platforms and/or unduly raise the costs of horizontal platform and downstream business user competitors. Instead of sanctioning firms retroactively for any harm they caused, rule-like presumptions seek proactively to avert anticompetitive harm from materializing, by making specific conduct unavailable or prohibitively costly for designated platforms.¹⁵²

The corollary of the modal character of legal rules or rule-like presumptions is that they encode the proposition that the benefits (costs) of preventively outlawing specific conduct across different possible worlds outweigh (fall short of) the benefits (costs) of a more minute, casuistic analysis that seeks to minimize type 1 errors resulting from over-inclusive and less differentiated rules, but at the same time is inherently prone to type 2 errors.¹⁵³ Rule-like presumptions thus encapsulate the premise that specific conduct of designated platforms tends to be in such an overwhelming number of cases anticompetitive or to produce anticompetitive harm of such an important order of magnitude that the costs of a more elaborate analysis of their actual effects are not outweighed by the presumptive gains flowing from the

^{150.} Advice of the Digital Markets Taskforce, supra note 3, para. 4.40.

^{151.} List, supra note 97, at 204-209.

^{152.} Advice of the Digital Markets Taskforce, supra note 3, paras. 4.33-4.34.

^{153.} Christiansen & Kerber, supra note 88, at 227-228.

reduction in type 1 error costs.¹⁵⁴ In other words, for specific conduct by powerful platforms the new platform regulations endorse a certain tolerance level of type 1 errors resulting from the inherently over-inclusive nature of legal rules.

What explains this shift from under-inclusive standards to over-inclusive rules or rule-like presumptions? A first reason for the adoption of broadly construed, self-executing rules is the recalibration of the error-cost framework. The important role of presumptions in platform regulations indeed signals that over- rather than under-enforcement is increasingly considered the optimal intervention standard in digital markets. It is a manifestation of the fundamental revision of the two axioms regarding the incidence of anticompetitive effects and costs/benefits of antitrust intervention underpinning the orthodox error-cost framework.

By codifying a number of theories of harm that formed part of recent antitrust cases and investigations against powerful online platforms, the rule-like presumptions encode the inference that certain forms of conduct by powerful platforms have a high probability of leading to anticompetitive outcomes. The increased weight of priors ([P(H) / P(B)]) in the assessment of platform conduct is hence testament to the reconsideration of the axiom underpinning the orthodox error-cost framework that the incidence of anticompetitive conduct tends to be low, that is [P(B) / P(H)] > 1 or conversely [P(H) / P(B)] < 1. If antitrust decision making starts from the belief encapsulated in the orthodox error-cost calculus that business conduct is most often pro-competitive or competitively neutral (i.e., [P(H) / P(B)] < 1), the assessment of additional case-specific evidence (i.e., [P(e|H) / P(e|B)]) must indicate a likelihood of anticompetitive effects of significant weight for the ratio of updated probability estimates to exceed 1 (so that $[P(B) / P(B)] \times [P(e|H) / P(e|B)] = [P(H|e) / P(B|e)] > 1$) and to warrant the finding of anticompetitive effects. Put differently, if antitrust decision making is predicated on the prior that conduct is more likely to be competitively beneficial than harmful, the factfinder has to establish a considerable amount of case-specific evidence that outweighs this presumption and tilts the balance in support of a finding of anticompetitive harm.

By revising the assumptions of the prior distribution of anti- and procompetitive effects for (specific) conduct of designated platforms and assuming that [P(B)/P(H)] < 1 or conversely [P(H)/P(B)] > 1, the recalibrated error-cost framework reduces the weight of case-specific evidence of anticompetitive effects (i.e., [P(e|H)/P(e|B)]) a factfinder needs to advance and consider for the updated ratio of probability estimate to exceed 1 and indicate anticompetitive harm. If one starts from the prior that specific conduct is on balance more likely to be anticompetitive than not (i.e., [P(H)/P(B)] > 1), little additional case-specific evidence (i.e., [P(e|H)/P(e|B)]) is needed to confirm the hypothesis of anticompetitive effects, that is, $[P(H)/P(B)] \times [P(e|H)/P(e|B)] = [P(H|e)/P(B|e)] > 1$. In turn, this also means that the stronger the presumption of anticompetitive effects, the more case-specific countervailing or off-setting evidence (i.e., [P(e|H)/P(e|B)]) must be proffered by designated platforms to rebut the presumption of anticompetitiveness and tilt the ratio of posterior probabilities toward non-intervention.¹⁵⁵

The new rule-like presumptions also embody the belief that specific conduct by designated platforms will result in harm of greater magnitude than the conduct of other smaller platforms or comparable non-digital firms. The greater weight of presumptions under the new platform regulations thus also marks the departure from the second axiom underpinning the orthodox error-cost framework that the costs of antitrust intervention tend to outweigh its benefits, that is [Loss / Gain] > 1 or conversely [Gain / Loss] < 1. Instead, the growing importance of legal presumptions gives currency to the view that anticompetitive harm in digital markets is not only more likely, but also often of greater magnitude and

^{154.} That is the classic rationale of per se rules under Section 1 Sherman Act. See Northern Pacific RY. Co. v. United States 356 US 1 (1958) 5.

^{155.} Kaplow, supra note 73, at 14.

more difficult to reverse than traditionally assumed by the orthodox error-cost framework.¹⁵⁶ Given the considerable magnitude of anticompetitive effects in digital markets (see discussion in section IV), the gains of intervention may outweigh its costs—that is [Loss / Gain] < 1 or conversely [Gain / Loss] > 1— even if the probability of the conduct to result in anticompetitive effects is low. The greater role of presumptions of anticompetitiveness in the new platform regulations is hence not only an expression of the greater probability of anticompetitive effects of specific platform conduct (reflected in the formalization of presumptions as prior probability estimates) but may also capture the size of their adverse consequences that are ignored by a purely probabilistic understanding of presumptions.

A second reason for the greater role of self-executing rule-like presumptions is that the new platform regulations seek to enhance legal certainty not only for regulated gatekeeper platforms but also for competing platforms, businesses, and end users.¹⁵⁷ Less differentiated legal commands, in the form of legal rules and rule-like presumptions, indeed, outperform standards in ensuring legal certainty and predictability for all relevant stakeholders.¹⁵⁸ This comparative advantage of legal rules over standards in guaranteeing legal certainty and predictability is of particular relevance in digital markets. Given that gatekeeper platforms perform a high volume of transactions, the new platform regulations are likely to govern transactions and conduct that occur with high frequency. The informational and compliance costs of rule-like presumptions for enforcers and individual firms are inferior to those of standards when the frequency of the governed conduct is high.¹⁵⁹ In securing greater legal certainty, rule-like legal presumptions may not only contribute to enhanced deterrence and greater compliance¹⁶⁰ but also secure the incentives of regulated platforms and other market participants alike to invest in innovation.

Third, as they obviate the need for market definition, the assessment of anticompetitive effects, and counter-factual analysis, rule-like presumptions reduce the amount of information competition authorities have to process and thereby facilitate swifter antitrust intervention. Rule-like presumptions thus allow affected consumers, competitors, or competition authorities to challenge anticompetitive conduct more readily and easily than open-textured standards.¹⁶¹ The adoption of self-executing rules is hence cognizant of the importance of the time-dimension of error costs in digital markets, which are prone to tipping. It also embodies the realization that modern antitrust law, owing to its reliance on standards and a casuistic effects-based analysis, has proven too slow in preventing anticompetitive conduct that tipped markets on a lasting basis in favor of a dominant platform.¹⁶²

The important role of legal presumptions in the new regulations of digital platforms also shows that a more economic approach must not necessarily call for more differentiated legal commands and inductive analysis. Once one departs from the axioms of the orthodox error-cost framework that the probability of anticompetitive harm is low and the costs of type 1 errors tend to exceed the benefits of antitrust intervention, a more deductive analysis and less differentiated legal commands may be consonant with an economically informed approach. The reliance of the new platform regulations on rule-like presumptions thus operationalizes the belief that in digital markets the costs of type 2 errors and complex

^{156.} Advice of the Digital Markets Taskforce, *supra* note 3, paras. 7, 14, 2.11, 3.11, 4.34; DCMS/BEIS Consultation Document, *supra* note 3, para. 69; Furman et al., *supra* note 39, paras. 3.89–3.91; Stigler Committee on Digital Platforms, *Final Report* (2019), 73–74. https://research.chicagobooth.edu/stigler/media/news/committee-on-digital-platforms-final-report.

^{157.} DMA Proposal, supra note 2, Explanatory Memorandum, at 5-6.

^{158.} Louis Kaplow, supra note 96, at 621; Christiansen & Kerber, supra note 88, at 219-220.

^{159.} Kaplow, supra note 96, at 564, 585.

John E. Calfee & Richard Craswell, Some Effects of Uncertainty on Compliance with Legal Standards, 75 VA LAW REV. 965 (1984). Richard Craswell & John E. Calfee, Deterrence and Uncertain Legal Standards, 2 J. LAW ECON. ORGAN. 279 (1986).

^{161.} Advice of the Digital Markets Taskforce, supra note 3, para. 4.34.

^{162.} Furman et al., *supra* note 39, paras. 2.8, 3.119.

competition law analysis (i.e., regulation costs) outweigh the accuracy benefits of greater differentiation in terms of lower type 1 errors.¹⁶³

This trade-off between error costs and accuracy benefits of antitrust intervention does not necessarily have to take the form of a strictly dichotomous choice between rules and standards. The DMA and the SMS regime also illustrate that a balance can be struck through the fine-tuning of the specific design and implementation of rule-like presumptions. For instance, arts. 5 and 6 DMA both set out rule-like presumptions of illegality for certain conduct of designated platforms. Art. 6, however, creates room for greater flexibility in the implementation of these rules, by according an important role to the Commission in further specifying and designing solutions and remedies at the individual platformlevel. Arts. 5 and 6 thus establish a framework for differentiated rules: while art. 5 presumptions uniformly apply to all gatekeepers across the board, art. 6 allows for more tailor-made interventions and remedies.¹⁶⁴ The proposed SMS regime goes one step further in creating wiggle room for further finetuning and differentiation of rule-like presumptions. While the codes of conduct envisaged by the SMS regime lay down, in a similar way to art. 5 DMA, specific rules that apply to all SMS platforms across the board, they provide at the same time the DMU with plenty of leeway to establish tailor-made rules for individual platforms.¹⁶⁵ By adopting a differentiated approach toward the design and fine-tuning of legal presumptions, the DMA and the SMS framework seek to minimize the type 1 errors associated with the inherent over-inclusiveness of legal rules by increasing the degree of differentiation of legal commands and blending rules with the flexibility of legal standards.

One might speculate about the extent to which the growing weight of presumptions in the new platform regulations and the heightened evidentiary burden for their rebuttal will also have spill-over effects on the competition law analysis of conduct or sectors not covered by the new platform regulations. This question is of particular relevance because the rule-like presumptions of the new platform regulations appear, at least in their current state, to primarily target non-price conduct, such as selfpreferencing, refusals to ensure interoperability, or tying. This focus on non-price conduct is somewhat puzzling in so far as price conduct, such as loyalty rebates or exclusivity payments by dominant platforms, often have similar foreclosure effects and allow platforms to leverage and entrench their market power.¹⁶⁶ No explanation is so far provided as to why most forms of exclusionary price conduct fall outside the scope of the new platform regulations. This is problematic not least because the disparate treatment of non-price and price conduct by dominant gatekeeper platforms is liable to lead to incoherent outcomes. By way of example, under the new platform regulations tying by a dominant gatekeeper platform will be presumed unlawful regardless of the efficiency of competitors. By contrast, loyalty rebates or exclusivity payments will continue to be subject to a case-by-case analysis and, under certain circumstances, the application of an as-efficient competitor test under Section 2 Sherman Act,¹⁶⁷ art. 102 TFEU¹⁶⁸ and/or equivalent provisions of national law (i.e. s. 19 GWB or the Chapter II prohibition of the Competition Act 1998), although they may have the very same exclusionary effects. This incoherence in treatment of exclusionary price and non-price conduct needs addressing. This does not mean that the presumptions of anticompetitiveness should be necessarily expanded to price conduct. There are indeed important reasons-for instance, the goal of preserving dominant platforms' incentives to compete aggressively on prices-that may support a dissimilar treatment of gatekeeper platforms' price and non-price conduct. However, the reasons underpinning this inconsistent treatment should be clearly articulated.

^{163.} Christiansen & Kerber, supra note 88, at 227-231.

^{164.} DMA Proposal, supra note 2, arts. 6 and 7 (2)-(7).

^{165.} Advice of the Digital Markets Taskforce, *supra* note 3, 4.33–4.44.

^{166.} Case No COMP/AT.40099 Google Android. C(2018) 4761 final paras. 1188-1332.

^{167.} Eisai, Inc. v. Sanofi Aventis U.S. LLC 821 F.3d 394 (2015) 408-409.

^{168.} Case C-413/14 P Intel v Commission ECLI:EU:C:2017:632 paras. 137-141.

VI. From Pure to Bounded Probabilism

A fifth element in which the new digital platform regulations depart from modern antitrust law relates to the standard of proof or threshold of intervention. Under the auspices of the more economic approach, conventional antitrust has increasingly moved toward what one can call "pure probabilism." Modern antitrust indeed largely endorsed a probabilistic standard of proof that requires competition law plain-tiffs to prove actual or likely anticompetitive effects for competition law intervention to be justified. The new platform regulations importantly digress from this purely probabilistic understanding of the standard of proof.

A. A Decision-Theoretic Operationalization of the Standard of Proof

The standard of proof is best understood as the critical quality threshold evidence has to meet for a specific fact—here antitrust harm H—to be considered true. Only if an antitrust plaintiff (i.e., a competition authority or private plaintiff) advances evidence that shows anticompetitive harm to this standard, will it discharge its initial burden of proof and antitrust intervention will be justified unless the defendant successfully puts forward countervailing evidence. Conventional literature tends to refer to the standard of proof as a critical likelihood value that posterior probabilities have to meet for antitrust harm H to be considered proven given the evidence e, that is P(H|e). The "balance of probabilities" or "preponderance of evidence" standard, which governs, for instance, U.S. antitrust law¹⁶⁹ and (phase II of) U.K. merger control,¹⁷⁰ requires that conduct be shown to be more likely than not to result in anticompetitive harm. For a plaintiff to discharge its initial burden under the balance of probabilities standard, it has to demonstrate that the posterior probability of H given evidence e slightly exceeds 50 percent, that is, P(H|e) > 0.5. Only in this case, P(H|e) is more likely than not; or in other words, more likely than the "null hypothesis" of the posterior probability of the conduct given evidence e being beneficial, that is P(B|e). Expressed as a critical likelihood ratio,¹⁷¹ the "balance of probabilities" or "preponderance of evidence" standard requires that the posterior odds ratio of the conduct being harmful given evidence e be greater than 1, that is [P(H|e)/P(B|e)] > [0.5/0.5] or 1.¹⁷²

What is, however, often ignored in the conventional discussion of the requisite standard of proof in antitrust law is that the purely probabilistic "balance of probabilities" or "preponderance of evidence" standard constitutes only in limited circumstances an optimal decision rule that minimizes error costs. A major shortcoming of this standard is that it only focuses on (posterior) probabilities of conduct to result in harm (or no-harm), without accounting for the consequences (that is, in economic parlance, the benefits and costs) of that harm (or no-harm).¹⁷³ Instead of merely focusing on probabilities, an optimal decision rule would also account for the benefits and costs of antitrust intervention. An optimal decision rule would thus counsel competition law (only) to prohibit conduct if its posterior probability of being harmful multiplied by the gains of preventing this harm (i.e., the scale of averted anticompetitive harm) exceeds the posterior probability of it being beneficial multiplied by the associated loss of mistakenly prohibiting such beneficial conduct (i.e., type 1 errors). Under this optimal decision rule, antitrust intervention is only warranted if the following condition is fulfilled:

$$P(H|e) \times Gain > P(B|e) \times Loss.$$

^{169.} C. Frederick Beckner III & Steven C. Salop, Decision Theory and Antitrust Rules, 67 ANTITRUST LAW J. 41, 61 (1999).

^{170.} Merger Assessment Guidelines 2021. CMA 129 para. 2.36.

^{171.} What follows draws on Kaplow, supra note 73, at 2-4, 6-10.

^{172.} Salop, supra note 90, at 14, fn 40.

^{173.} Beckner and Salop, supra note 169, at 61-62; Kaplow, supra note 73, at 18.

This can also be expressed as an odds ratio:

$$\frac{P(H|e)}{P(B|e)} > \frac{Loss}{Gain}.$$

If we replace the posterior probabilities ratio [P(H|e)/P(B|e)] by Bayes' theorem (i.e., $[P(H|e)/P(B|e)] = [P(H)/P(B)] \times [P(e|H)/P(e|B)]$), the optimal rule would suggest that antitrust intervention is only desirable if

$$\frac{P(H)}{P(B)} \times \frac{P(e|H)}{P(e|B)} > \frac{Loss}{Gain} \text{ or if } \frac{P(e|H)}{P(e|B)} > \frac{P(B)}{P(H)} \times \frac{Loss}{Gain}$$

Instead of associating the standard of proof with a critical ratio of posterior probabilities [P(H|e)/P(B|e)], decision theory, therefore, suggests that the (optimal) standard of proof should be defined in relation to the likelihood ratio (LR) of the conditional probability that evidence *e* is associated with anticompetitive harm P(e|H) relative to the conditional probability that the same evidence is observed with procompetitive benefits P(e|B), that is [P(e|H)/P(e|B)].¹⁷⁴

The difference between the optimal decision rule (i.e., $[P(e|H)/P(e|B)] > [P(B)/P(H)] \times [Loss/Gain]$) and the balance of probabilities standard (i.e. [P(H|e)/P(B|e)] > 1) is striking. The optimal decision rule suggests that the standard of proof should be calibrated to both the prior like-lihood (i.e., [P(B)/P(H)]) and scale of antitcompetitive effects (i.e. [Loss/Gain]). The balance of probabilities standard, meanwhile, militates for the static application of the very same critical like-lihood ratio ([0.5/0.5] or 1) across all types of cases, regardless of the prior probability and consequences of specific conduct resulting in anticompetitive harm.

Contrasting the balance of probabilities standard with the optimal decision rule is insightful because it reveals the extent to which the balance of probabilities standard encodes the two fundamental axioms that underpin the orthodox error-cost framework. The optimal decision rule and the balance of probabilities standard, in fact, only coincide if the ratio of benefits and losses of antitrust intervention (i.e., [Loss / Gain] > 1) and prior odds ratio of anticompetitive benefits and harm (i.e., [P(B) / P(H)] > 1) both exceed 1. Only if this is the case, $[P(e|H) / P(e|B)] > [P(B) / P(H)] \times [Loss / Gain]$ equals [P(e|H) / P(e|B)] > 1.¹⁷⁵

This shows that the balance of probabilities standard does nothing else than consolidating the two basic assumptions underpinning the orthodox error-cost framework: namely, that firm conduct is more likely than not to be pro-competitive or neutral (i.e., [P(B)/P(H)] > 1) and that the costs of antitrust intervention tend to exceed its benefits (i.e. [Loss/Gain] > 1). The balance of probabilities standard hence encodes the implicit default assumption that firm conduct is on balance marginally pro-competitive or competitively neutral with ties going to the defendant.¹⁷⁶

While U.S. antitrust law is largely aligned with this balance of probabilities standard, E.U. competition law and notably art. 102 TFEU for a long time did not follow its purely probabilistic logic. Instead, the E.U. Courts consistently held that for unilateral conduct to be caught by art. 102 TFEU the Commission was under no obligation to demonstrate that it entailed actual or likely anticompetitive effects. Rather, it was deemed sufficient that the Commission advances evidence showing that the conduct was capable of causing potential anticompetitive effects.¹⁷⁷ In contrast to the balance of

^{174.} Kaplow, supra note 73; Louis Kaplow, Burden of Proof, 121 YALE LAW J. 738, 772-786 (2012).

^{175.} For a similar point, Burtis et al., supra note 148, at 14-15.

^{176.} Salop, supra note 90, at 6, 14 fn. 40.

^{177.} Case 322/81 Michelin v Commission ECLI:EU:C:1983:313 paras. 73, 81. Case C-95/04 P British Airways plc v Commission of the European Communities ECLI:EU:C:2007:166 paras. 67–68. The same standard also governed Art 101 TFEU, see for instance Case C-8/08 T-Mobile Netherlands BV and Others ECLI:EU:C:2009:343 para. 31.

probabilities standard, this "capability standard"¹⁷⁸ is probabilistically de-weighted as it does not make the proof of anticompetitive effects conditional on their posterior probabilities being shown to be more likely than not, that is a likelihood in excess of 50 percent. With the rise of the more economic approach, this capability standard of proof has become the target of continuous criticism.¹⁷⁹ Proponents of the more economic approach perceived it as a major obstacle to a more effects-based analysis that would require the Commission to ascertain the actual or likely effects of the impugned conduct. The Commission reacted to this criticism by endorsing a balance of probabilities standard for the assessment of coordinated and unilateral conduct, as well as mergers, as part of its modernization efforts.¹⁸⁰ The E.U. Courts, however, failed to fully align E.U. competition law with the balance of probabilities standard. So far, they only endorsed the balance of probabilities standard in the realm of merger control.¹⁸¹ Recently, the General Court went even further, asserting that at least for some mergers the standard of proof may be stricter than the balance of probabilities standard.¹⁸² Under art. 102 TFEU, the Court of Justice also seemed to incrementally depart from the capability standard and replace it with the balance of probabilities standard.¹⁸³ In Post Danmark II, the Court, for instance, held that the anticompetitive effects must be shown to be, if not actual, at least "likely"¹⁸⁴ or "probable."¹⁸⁵ More recent cases, such as Intel, however, seem to revert back to the capability standard.¹⁸⁶

B. The Recalibration of the Standard of Proof

The new platform regulations dent the gravitation of modern U.S. and European antitrust toward a probabilistic standard of proof. All four platform regulations in the E.U., the U.S., the U.K. and Germany seem to propose a partial recalibration of the standard of proof in favor of a probabilistically de-weighted standard that gives greater weight to the magnitude of anticompetitive harm alongside its probability.

Take, for instance, the revised German Competition Law 4.0. For the presumptions of anticompetitiveness of specific types of conduct in s. 19a (2) to become operative, no showing of actual or likely anticompetitive effects is necessary. Rather, these presumptions encode an inference of the increased "potential" (*"Schädigungspotential*") of specific forms of conduct by firms holding paramount significance for competition to cause material harm.¹⁸⁷ Similarly, the draft DMA disavows recent attempts to align E.U. competition law with a purely probabilistic standard of proof. The proposal makes it clear that blacklisted conduct by gatekeeper firms does not have to be shown to result in "actual, likely or presumed effects" for it to be prohibited.¹⁸⁸

181. Case C-413/06 P Bertelsmann and Sony Corporation of America v Impala ECLI:EU:C:2008:392 para. 47.

^{178.} Pablo Ibáñez Colomo & Alfonso Lamadrid de Pablo, On the Notion of Restriction of Competition: What We Know and What We Don't Know We Know, in The Notion of Restriction of Competition: Revisiting the Foundations of Antitrust ENFORCEMENT IN EUROPE 361–63 (D. Gerard et al., ed. 2017).

^{179.} See for instance John Kallaugher & Brian Sher, Rebates Revisited, 25 Eur. COMPET. LAW REV. 263 (2004).

^{180.} See for instance for Art. 102 TFEU, Guidance on the Commission's enforcement priorities in applying Article 82 of the EC Treaty to abusive exclusionary conduct by dominant undertakings. OJ [2009] C 45/7 paras. 19–20.

^{182.} Case T-399/16 CK Telecoms UK Investments v Commission, supra note 61, para. 118.

^{183.} Opinion of Advocate General Kokott in Case C-23/14 Post Danmark II ECLI:EU:C:2015:343 para. 82.

^{184.} Case C-23/14 Post Danmark II ECLI:EU:C:2015:651 paras. 67, 69. See also Case C-209/10 Post Danmark A/S v Konkurrencerådet, supra note 59, para. 44.

^{185.} Case C-23/14 Post Danmark II, supra note 184, para. 74 and operative part.

^{186.} Case C-413/14 P Intel v Commission, supra note 168, paras. 138, 140; Case C-307/18 Generics (UK) and Others ECLI:EU:C:2020:52 para. 154; Case T-612/17 Google and Alphabet v Commission (Google Shopping), supra note 59, para. 438.

Legislative proposal of the Competition Law 4.0, *supra* note 57, 73, 76; Parliamentary Resolution on the Competition Law 4.0, *supra* note 23, at 113–115. Franck and Peitz, *supra* note 10, at 520.

^{188.} DMA Proposal, supra note 2, recitals 9-10.

A similar lowering of the standard of proof is also contemplated by the proposed U.K. SMS regime and the U.S. platform regulations. In a similar way to their E.U. and German counterparts, they outlaw certain unilateral platform conduct without requiring the showing of actual or likely anticompetitive effects.¹⁸⁹ The recalibration of the standard of proof under both regimes is even more marked in relation to platform acquisitions. Whereas the DMA only provides for a reporting obligation for mergers by gatekeeper platforms regardless of whether they hit the jurisdictional thresholds of the E.U. Merger Regulation 139/2004,¹⁹⁰ the U.K. SMS and U.S. regulations contain specific substantive provisions for the assessment of platform acquisitions.

These new rules on platform mergers seek to address concerns over rising levels of start-up acquisitions by powerful incumbent platforms, most of which escaped conventional merger scrutiny. Commentators have grown fearful that these start-up acquisitions enabled incumbent platforms to eliminate potential competitive threats to their market positions either by integrating them into their ecosystem or by discontinuing their innovation projects. Start-up acquisitions by powerful incumbent platforms are, therefore, increasingly viewed as a threat to innovation in the digital economy.¹⁹¹ At the same time, startup acquisitions pose a major challenge for competition authorities. They require them to make a judgment on the impact of mergers on competition and innovation which might only be felt in a relatively distant future. To scrutinise these mergers, competition authorities need to make long-term predictions about competitive effects which exceed the usual 2- to 5-year timeframe of standard merger analysis. As the impact of these mergers on competition and innovation is largely shrouded in uncertainty, it is virtually impossible to make a probabilistic judgment about whether, on the balance of probabilities, such acquisitions are more likely than not to cause adverse effects on future competition and innovation.

For this reason, the U.K. SMS regime proposes to vet start-up acquisitions by platforms holding SMS status under a lower standard of proof than the one that governs ordinary merger review. Rather than being required to show that a platform merger will result on a balance of probabilities in anticompetitive effects, it will be sufficient for the CMA to demonstrate that it gives rise to a "realistic prospect" of a SLC to challenge it in phase 2 of the merger review.¹⁹² Under this lower and more cautious "realistic prospect" standard of proof which is normally reserved to the phase 1 assessment of "conventional" U.K. merger analysis, the CMA will be able to challenge and block platform acquisitions whose anticompetitive effects will materialize with a probability of less than 50 percent. This lowering of the standard of proof opens up new possibilities for the CMA to intervene against transactions which, owing to the size of the acquirer and/or the innovative capacities of the target, have the potential to cause large-scale, but low-probability harm.

The U.S. proposals go one step further than the U.K. SMS regime in lowering the standard of proof for the assessment of start-up acquisitions. By introducing a prohibition of platform acquisitions, save the showing of their innocuous anticompetitive impact by the designated platform, the U.S. proposals create a presumption of anticompetitiveness against platform acquisitions.¹⁹³ The implicit consequence of this presumption is that the competition authorities' standard of proof for the assessment of start-up acquisitions is practically reduced to close to 0. By de facto reversing the burden of proof, this presumption empowers the Federal Trade Commission and Department of Justice to challenge platform mergers without proving that they will be likely to engender anticompetitive effects.

The new platform regulations thus endorse a lowered, probabilistically de-weighted standard of proof that is alert to the mere potential of specific conduct to cause anticompetitive harm. Instead of focusing

^{189.} ACIO Act, supra note 4, s. 2 (a) and (b).

^{190.} DMA Proposal, supra note 2, art. 12.

^{191.} Furman et al., supra note 39, paras. 3.42–3.108; Crémer et al., supra note 39, at 110–24; OECD, Start-Ups, Killer Acquisitions and Merger Control—Background Note (2020), https://one.oecd.org/document/DAF/COMP(2020)5/en/pdf.

^{192.} Advice of the Digital Markets Taskforce, supra note 3, points 4.149–4.153 and Annex F points 89–123.

^{193.} PCO Act, supra note 4, s. 2 (a) and (b).

exclusively on the probability of certain conduct by designated platforms to result in harm, this approach is rather concerned about the magnitude of harm that it may cause. This probabilistically de-weighted standard of proof thus allows for intervention against conduct or mergers that have the potential to cause large-scale harm, without this harm necessarily being more likely than not to materialize.

C. The Rationale Underpinning the Recalibration of the Standard of Proof

In a similar vein as the greater reliance on presumptions, this recalibration of the standard of proof is the immediate result of the reconfiguration of the error-cost framework underpinning the new platform regulations. The new platform regulations overturn the basic axioms of the orthodox error-cost framework, that conduct tends to be most of the time pro-competitive (i.e., [P(B)/P(H)] > 1) and that the costs of antitrust intervention usually outweigh its benefits (i.e., [Loss/Gain] > 1). Rather they ground in the assumption that certain platform conduct has a greater probability of resulting in harm than benefits (i.e., [P(B)/P(H)] < 1) and/or that the scale of this harm is so large that the benefits of antitrust intervention outweigh its costs in terms of false positives (i.e., [Loss/Gain] < 1). In such a setting, decision theory suggests that the optimal standard of proof may lie below the balance of probabilities standard. This lower standard directly follows from the optimal decision-theoretic rule $[P(e|H)/P(e|B)] > [P(B)/P(H)] \times [Loss/Gain]$ which posits that the error-cost framework and standard of proof should not only account for the probability of anticompetitive harm, but also factor in the magnitude or scale of this harm.¹⁹⁴

This lowered standard of proof is thus in keeping with decision theory which counsels against a "uniform" and "static" standard of proof, such as the "balance of probabilities" standard. Rather, it is consistent with the insight that the standard of proof should be commensurate to both the likelihood and magnitude of the potential harm caused by anticompetitive conduct or mergers. Accordingly, if specific conduct or mergers may result in anticompetitive effects with (1) a high probability and/or (2) a high magnitude of harm, decision theory would support the standard of proof to be set below the "balance of probabilities" standard.¹⁹⁵ Recent expert reports suggest that such a lowering of the standard of proof is particularly warranted in digital markets where the conduct of an incumbent firm may contribute to future market tipping and result in low-probability but high-impact harm.¹⁹⁶

The shift toward a probabilistically de-weighted standard of proof is also cognizant of the fact that competition authorities often have to take decisions under radical uncertainty.¹⁹⁷ Consistent with the work of Chicago School economists,¹⁹⁸ the conventional antitrust literature tends to assume that any type of uncertainty about future events can be expressed and assessed by factfinders as subjective probability estimates or discounted expected utilities.¹⁹⁹ This assumption however obfuscates the seminal distinction coined by the economist Frank Knight between "risk" and "uncertainty."²⁰⁰ Knight highlighted that risk, which refers to "measurable uncertainty" that can be captured by assigning probabilities to specific events or outcomes, fundamentally differs from (non-measurable) uncertainty to

^{194.} Beckner & Salop, *supra* note 169, at 61-62; Kaplow, *supra* note 73, at 18-20.

Louis Kaplow, *Efficiencies in Merger Analysis*, Harvard M. Olin Center for Law, Economics and Business Policy Discussion Paper 3/2021 (2021), 46–47, http://www.law.harvard.edu/programs/olin_center/papers/pdf/Kaplow_1056.pdf.
 Furman et al., *supra* note 39, paras. 3.88–3.100. Crémer et al., *supra* note 39, at 50–51.

^{197.} For the concept of radical uncertainty, see John Kay & Mervyn King, Radical Uncertainty: Decision-making for an unknowable future (Little, Brown 2020).

Milton Friedman & Leonard J. Savage, *The Expected-Utility Hypothesis and the Measurability of Utility*, 60 J. POLIT. ECON. 463, 467 (1952).

^{199.} Michael L. Katz & Howard A. Shelanski, *Merger Analysis and the Treatment of Uncertainty*, 74 ANTITRUST LAW J. 537 (2007).

^{200.} FRANK KNIGHT, RISK, UNCERTAINTY AND PROFIT (Houghton Mifflin Harcourt 1921), 19-20.

| Parameters | Modern Antitrust | New Platform Regulations |
|--|---|--|
| Goals Error-cost framework | Consumer welfare standard $\frac{P(B)}{P(H)} > 1$ or conversely $\frac{P(H)}{P(B)} < 1$ $\frac{Loss}{Gain} > 1$ or conversely $\frac{Gain}{Loss} < 1$ | $\begin{array}{l} \mbox{Multi-value approach} \\ \mbox{$\frac{P(B)}{P(H)} < 1$ or conversely $\frac{P(H)}{P(B)} > 1$} \\ \mbox{$\frac{Loss}{Gain} < 1$ or conversely $\frac{Gain}{Loss} > 1$} \end{array}$ |
| Mode of decision making and form of legal commands | Inductive Legal standards $\downarrow \frac{P(H)}{P(B)} \times \uparrow \frac{P(e H)}{P(e B)} = \frac{P(H e)}{P(B e)}$ | Deductive Rule-like presumptions $\uparrow \frac{P(H)}{P(B)} \times \downarrow \frac{P(e H)}{P(e B)} = \frac{P(H e)}{P(B e)}$ |
| Standard of proof | Probabilistic Balance of probabilities or preponderance of evidence $\frac{P(H e)}{P(B e)} > I$ | Probabilistically de-weighted Calibrated to both the probability and size of anticompetitive harm $\frac{P(e H)}{P(e B)} > \frac{P(B)}{P(H)} \times \frac{Loss}{Gain}$ |

Table 6. The Core Parameters of the Reconfiguration of Antitrust.

which no probabilistic value can be attributed. Uncertainty (in the strict, Knightian sense) about the unknown future development of markets prevents competition authorities from assigning probabilities to various multiple scenarios and to pick, in keeping with the balance of probabilities standard, the most likely one. Such unmeasurable uncertainty is notably present in fast-moving innovationdriven markets where the future development of competition and innovation is shrouded in uncertainty. By moving away from the balance of probabilities standard toward a probabilistically de-weighted standard that is primarily concerned about the potential harm certain conduct may generate, the recalibration of the standard of proof in the new platform regulations realigns antitrust law with the fundamental Knightian distinction between risk and uncertainty which has been glossed over for far too long by the conventional antitrust analysis.

VII. Conclusion—Toward a New Paradigm of Innovation

This article reflects on the various ways in which the new initiatives to regulate powerful online platforms in the European Union, the United States, the United Kingdom, and Germany challenge wellestablished fundamentals of modern antitrust and thereby reshape the future of competition law. The paper shows that the new platform regulations set in motion a profound transformation of modern antitrust law that operates along four parameters (summarized in Table 6).

First, the new platform regulations unsettle the long-standing baseline assumption that the maximization of consumer welfare constitutes competition law's core mission. Pursuing the goals of inter alia greater contestability, fairness, innovation, and choice, the new platform regulations replace a narrow understanding of the consumer welfare standard with a multi-value approach that seeks to address the multi-dimensional economic and social consequences of entrenched economic power in digital markets.

Second, the new platform regulations repudiate the orthodox understanding of error-costs that advocates under-enforcement as the optimal standard of intervention, notably, in digital markets. This orthodox error-cost framework revolved around the twin axioms that the probability of anticompetitive

effects is low ([P(B)/P(H)] > 1 or [P(H)/P(B)] < 1) and that the costs of antitrust intervention tend to outweigh its benefits ([Loss/Gain] > 1 or [Gain/Loss] < 1). The new platform regulations embody a recalibrated error-cost framework that overturns both axioms. The revised error-cost framework asserts that the conduct of powerful platforms is not only more likely to result in anticompetitive harm than conventional antitrust literature assumes (i.e., [P(B)/P(H)] < 1 or [P(H)/P(B)] > 1), but also posits that the harm averted by accurate antitrust enforcement is of such significant magnitude that the gains of intervention tend to exceed its costs (i.e. [Loss/Gain] < 1 or [Gain/Loss] > 1). On the basis of these revised assumptions about the probability distribution and magnitude of anticompetitive harm, over- rather than under-enforcement is considered the optimal policy standard in digital markets.

Third, the new platform regulations give effect to this recalibrated error-cost framework by partially reversing the trend toward an increasingly inductive mode of analysis that characterized modern antitrust under the "more economic" or "effects-based" approach. This inductive mode disavowed priors, in the form of legal rules and presumptions (i.e., [P(H)/P(B)]), in favor of a casuistic analysis of case-specific evidence (i.e., [P(e|H)/P(e|B)]) and highly differentiated legal standards ($\downarrow [P(H)/P(B)] \times \uparrow [P(e|H)/P(e|B)] = [P(H|e)/P(B|e)]$). By opting primarily for rule-like legal presumptions as legal commands to regulate competition in digital markets, the new platform regulations resurrect a deductive mode of analysis that enables enforcers to rely on strong priors to infer the anticompetitive nature of platform conduct from its legal or economic form ($\uparrow [P(H)/P(B)] \times \downarrow [P(e|H)/P(e|B)] = [P(H|e)/P(B|e)]$).

Fourth, the recalibrated error-cost framework underpinning the new platform regulations also finds its expression in the lowering of the standard of proof for finding anticompetitive conduct. The new platform regulations fundamentally diverge from the purely probabilistic "balance of probabilities" or "preponderance of evidence" standard that governs conventional antitrust cases and is favored by proponents of an effects-based approach. This probabilistic standard requires antitrust plaintiffs to advance evidence showing that the impugned conduct would be more likely than not—that is, have a probability of above 50 percent—to engender anticompetitive harm (i.e., [P(H|e)/P(B|e)] > 1). Instead, the new platform regulations endorse a probabilistically de-weighted standard of proof that accounts for both the (1) probability and (2) magnitude of antitrust harm (i.e., $[P(e|H)/P(e|B)] > [P(B)/P(H)] \times [Loss/Gain)]$). This lowered standard of proof empowers competition authorities to intervene against conduct that has the mere potential of resulting in low-probability but high-impact harm.

This reconfiguration the new platform regulations are set to bring about in relation to the goals and modus operandi of modern antitrust have far-reaching implications. The transformation of the core parameters of modern antitrust law by the new platform regulations is, in fact, suggestive of a partial rethinking of the type of innovation competition law is supposed to protect in digital markets. In recalibrating the error-cost framework, legal commands, and standard of proof, the new platform regulations implicitly discard the Schumpeterian conception of innovation²⁰¹ that for a long-time shaped mainstream antitrust enforcement, most notably in dynamic, high-tech markets. Central to this Schumpeterian understanding of innovation is the concern about the ability of large-scale incumbents to appropriate and recoup their investments in the development of innovative technology, products, and services. On this account, immediate harm caused by market concentration and monopoly power is condoned in the name of future, albeit uncertain, benefits generated by greater innovation on the part of powerful firms.²⁰² This hands-off approach toward monopoly power grounds in the assumption that without having the possibility of appropriating their investments in innovation through the exercise of market power, large firms would not have the incentive to innovate in the first place.²⁰³ By virtue of its Schumpeterian

^{201.} SCHUMPETER, supra note 77, Chapter VII.

^{202.} Verizon Communications Inc v Law Offices of Curtis Trinko 540 US 398 (2004) 407.

underpinning, the conventional antitrust approach toward innovation thus almost exclusively focuses on the incentives of powerful incumbents to innovate. The impact of concentrated market power on the incentives of smaller rivals and future challengers to innovate is largely extraneous to this analysis.

The new platform regulations significantly depart from this one-sided reading of Schumpeterian innovation that underpins orthodox accounts of the role antitrust law should play in innovation-driven markets. Rather than being exclusively concerned about the incentives of incumbent firms to innovate, they shift the focus on the ability of smaller firms to appropriate their investment in follow-on and disruptive innovation. The new platform regulations recognize that the prospect of powerful platforms able to exclude smaller players from the market and to appropriate their investment in innovation at will, combined with the insight that they might be long out of business before antitrust law remedies anticompetitive conduct, may deter smaller players from innovating. As a result, innovation efforts and diversity may be irreversibly lost.²⁰⁴

By seeking to ensure a level playing field and contestability in digital markets, the new platform regulations explicitly aim to preserve smaller business users' and rivals' sunk investments in digital follow-on or disruptive innovation.²⁰⁵ The importance new platform regulations assign to the protection of a level playing field and equal competitive opportunities resonates well with economic research that documents the historical importance of inclusive economic institutions characterized by broad-based economic opportunities for economic progress and innovation.²⁰⁶ The reliance of the new platform regulations on legal presumptions and a probabilistically de-weighted standard of proof also contributes to the preservation of the incentives of smaller firms to innovate. Owing to their modal character, the legal presumptions and probabilistically de-weighted standard of proof afford smaller innovators with a resilient protection against arbitrary interference by powerful platforms not only in the present or likely neighboring but across a range of relevant possible worlds.

The inclusive model of innovation and growth envisaged by the new platform regulations becomes most apparent in the fact that they protect competing platforms and business users regardless of their efficiency. The new regulatory frameworks for digital markets are hence not exclusively directed against welfare-reducing conduct that reduces total/consumer welfare by foreclosing efficient competitors offering products/services that are not supplied by powerful digital platforms at all, or at less favorable conditions. Instead, they protect the revenue streams of competitors regardless of their efficiency. The new platform regulations are not only wary about excessive surplus extraction on the part of powerful platforms, but also epitomise a concern about a fair distribution of the surplus created by the rise of digital technologies. The new digital regulations thus incorporate important decisions about the distribution of power of rent extraction along digital value chains and pay special attention to processes of value generation and capture. In this regard, platform regulations are suggestive of a new model of economic growth in the digital economy, grounded in a broad-based distribution of innovative and economic opportunities.

Critics of the new platform regulations may object that this novel, more inclusive vision of innovation and growth constitutes an undue attempt to "politicise" antitrust law and to inject arbitrariness and value judgments into what they perceive as a depoliticised neutral and rational antitrust enterprise in which expertise primes particular interests, politics, and populism.²⁰⁷ That criticism is,

^{204.} Arrow & Fisher, supra note 85, at 319; Joseph Farrell, Complexity, Diversity, and Antitrust, 51 ANTITRUST BULL. 165 (2006).

^{205.} Carl Shapiro, *Competition and Innovation: Did Arrow Hit the Bull's Eye?*, in THE RATE AND DIRECTION OF INVENTIVE ACTIVITY REVISITED 400–401 (J. Lerner & S. Stern, ed. 2012), https://www.nber.org/books-and-chapters/rate-and-direction-inventive-activity-revisited/competition-and-innovation-did-arrow-hit-bulls-eye.

^{206.} DARON ACEMOGLU & JAMES A. ROBINSON, WHY NATIONS FAIL: THE ORIGINS OF POWER, PROSPERITY, AND POVERTY 76–77, 208, 319–34 (Crown Publishing 2013).

^{207.} Douglas A. Melamed & Nicolas Petit, The Misguided Assault on the Consumer Welfare Standard in the Age of Platform Markets, 54 Rev. IND. ORGAN. 741, 746, 767 (2019); Nicolas Petit, A Theory of Antitrust Limits, 28 Geo. MASON L. Rev. 1399, 1419 (2020).

however, unconvincing in two respects. First, its pretense to scientific truth or expertise fails to acknowledge that a more "precautionary" approach toward type 2 errors can be fully consistent with decision theory, as long as one relaxes some of the strong axioms on which the orthodox error-cost framework is predicated. Second, its pretense to apolitical neutrality is oblivious to the fact that the orthodox error-cost framework is based on strong assumptions that, themselves, encode a value judgment that consistently favors the interests of incumbent "haves" to the detriment of the new-coming "have nots." The orthodox error-cost framework and the underpinning Schumpeterian vision of innovation trade off short-term inefficiencies in the name of the prospect of future, albeit uncertain, economic progress.

The revised understanding of innovation underpinning the new platform regulations only differs from the value judgments incorporated in the orthodox error-cost framework in so far as it strikes a different bargain. On one hand, the new platform regulations are less willing to condone short-term harm resulting from anticompetitive conduct by powerful platforms to preserve future, yet uncertain, gains of dynamic efficiencies. On the other hand, in the case of doubt, they are prepared to tolerate losses in short-term efficiencies or dynamic efficiencies generated by a few large platforms in light of future broad-based innovation by multiple smaller players. So long as this value judgment is clearly articulated, and its costs are assumed and remain contestable, this inclusive notion of innovation is not more objectionable or irrational than the Schumpeterian one that fashioned the approach of conventional antitrust toward innovation. On the contrary, if fully and openly assumed, it has the merit of being more transparent than the value judgment obfuscated by the allegedly "neutral" and "depoliticised" orthodox error-cost framework which has an implicit, in-built bias against antitrust intervention.

Authors Note

The author is grateful to Stavros Makris, Morten Hviid, and Ben Evans for their insightful comments on previous versions of this article.

Declaration of Conflicting Interests

The author(s) declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Funding

The author(s) received no financial support for the research, authorship, and/or publication of this article.